Contents lists available at SciVerse ScienceDirect

Earth-Science Reviews

journal homepage: www.elsevier.com/locate/earscirev

Current research issues related to post-wildfire runoff and erosion processes

John A. Moody ^{a,*}, Richard A. Shakesby ^b, Peter R. Robichaud ^c, Susan H. Cannon ^d, Deborah A. Martin ^a

^a National Research Program, U.S. Geological Survey, Boulder, CO, USA

^b Department of Geography, College of Science, Swansea University, Wales, UK

^c Rocky Mountain Research Station, U.S. Department of Agriculture, Forest Service, Moscow, ID, USA

^d Landslide Hazards Program, U.S. Geological Survey, Golden, CO, USA

ARTICLE INFO

Article history: Received 11 April 2012 Accepted 11 March 2013 Available online 27 March 2013

Keywords: Wildfire Burn severity Meso-scale rainfall Soil-hydraulic properties Runoff Soil erosion

ABSTRACT

Research into post-wildfire effects began in the United States more than 70 years ago and only later extended to other parts of the world. Post-wildfire responses are typically transient, episodic, variable in space and time, dependent on thresholds, and involve multiple processes measured by different methods. These characteristics tend to hinder research progress, but the large empirical knowledge base amassed in different regions of the world suggests that it should now be possible to synthesize the data and make a substantial improvement in the understanding of post-wildfire runoff and erosion response. Thus, it is important to identify and prioritize the research issues related to post-wildfire runoff and erosion. Priority research issues are the need to: (1) organize and synthesize similarities and differences in post-wildfire responses between different fire-prone regions of the world in order to determine common patterns and generalities that can explain cause and effect relations; (2) identify and quantify functional relations between metrics of fire effects and soil hydraulic properties that will better represent the dynamic and transient conditions after a wildfire; (3) determine the interaction between burned landscapes and temporally and spatially variable meso-scale precipitation, which is often the primary driver of post-wildfire runoff and erosion responses; (4) determine functional relations between precipitation, basin morphology, runoff connectivity, contributing area, surface roughness, depression storage, and soil characteristics required to predict the timing, magnitudes, and duration of floods and debris flows from ungaged burned basins; and (5) develop standard measurement methods that will ensure the collection of uniform and comparable runoff and erosion data. Resolution of these issues will help to improve conceptual and computer models of post-wildfire runoff and erosion processes.

Published by Elsevier B.V.

Contents

1.	Intro	duction	11
2.	Frame	ework for organizing post-wildfire runoff and erosion responses	12
	2.1.	Fire regimes	12
	2.2.	Precipitation regimes	12
	2.3.	Hydro-geomorphic regime	13
	2.4.	Post-wildfire response domains	13
3.	Precip	pitation: research issues related to the temporal and spatial variability of meso-scale precipitation	13
	3.1.	Background	13
	3.2.	Temporal variations in precipitation	14
	3.3.	Spatial variations in precipitation	15
	3.4.	Future research priorities	16
4.	Infiltr	ation: quantifying soil properties and hydraulic effects	17
	4.1.	Background	17
	4.2.	Burn severity and soil hydraulic properties	19
	4.3.	Modification of traditional infiltration theories	20
	4.4.	Measurements of soil hydraulic properties	20

* Corresponding author. Tel.: +1 303 541 3011.

E-mail addresses: jamoody@usgs.gov (J.A. Moody), r.a.shakesby@swansea.ac.uk (R.A. Shakesby), probichaud@fs.fed.us (P.R. Robichaud), cannon@usgs.gov (S.H. Cannon), damartin@usgs.gov (D.A. Martin).

0012-8252/\$ – see front matter. Published by Elsevier B.V. http://dx.doi.org/10.1016/j.earscirev.2013.03.004







	4.5.	Time dependence issue	21
	4.6.	Future research directions	21
5.	Runo	\mathcal{M} inking precipitation and basin morphology to post-wildfire response \ldots	21
	5.1.	Background	21
	5.2.	Factors affecting post-wildfire runoff	22
		5.2.1. Critical thresholds.	22
		5.2.2. Contributing area	22
		5.2.3. Depression storage and surface roughness	23
		5.2.4. Connectivity of overland-flow generating areas	23
	5.3.	The significance of geomorphic properties	24
	5.4.	Future research directions	24
6.	Soil a	and sediment erosion and transport	24
	6.1.	Background	24
		6.1.1. Dry ravel	25
		6.1.2. Raindrop, rain-flow, and interrill erosion	25
		6.1.3. Rill erosion	26
		6.1.4. Landslides and slope failures	26
		6.1.5. Drainage, channel, and debris flow erosion	26
	6.2.	Sediment availability	27
	6.3.	Post-wildfire erosion and transport measurements	27
	6.4.	Causes of differences in processes between post-wildfire domains	28
	6.5.	Future research directions	29
7.	Sumn	mary	29
Ack	nowled	dgments	30
Refe	s	30	

1. Introduction

The number and severity of wildfires in the United States and in other parts of the world have become a major concern in recent decades. This partly stems from second-order impacts and concerns about carbon storage, water quality, and ecosystem disturbance, but mostly from concerns related to the increases in population in or near wildfire-prone areas where post-wildfire enhanced runoff and erosion can result in catastrophic damage and loss of life by destructive floods and debris flows (Neary and Gottfried, 2002; Pausas et al., 2008). Post-wildfire responses tend to be disproportionately large compared to the size of the burned basin. For example, peak discharges can be as large as 300 $\text{m}^3 \text{ s}^{-1} \text{ km}^{-2}$ (Brown, 1972; Moody and Martin, 2001a; Gartner et al., 2004; Moody et al., 2008a; Smith et al., 2011a), which is comparable to and in some cases greater than the maximum rainfallrunoff floods in unburned conditions (Costa, 1987). Furthermore, the impacts of climate change on wildfire ignitions and behavior have been actively researched for some time (Flannigan et al., 2000; Westerling et al., 2003; Bachelet et al., 2007; Littell et al., 2009; Moritz et al., 2010; Westerling et al., 2011), but the implications for post-wildfire runoff and erosion are only now being explored (Pierce and Meyer, 2008; Moody and Martin, 2009a; Goode et al., 2012).

Continued progress in understanding and predicting post-wildfire runoff and erosion processes is hindered by a number of limitations. First, the responses of burned areas are transient, often lasting less than 7 years, depending on various aspects, notably the speed of vegetation recovery, post-wildfire weather conditions, sediment availability, and basin morphology (Rowe et al., 1954; Cerdà, 1998a; Moody and Martin, 2001b; Gartner et al., 2004; Shakesby et al., 2007; Sheridan et al., 2007; Cannon et al., 2010). Such a transient response limits the duration of the research window (Moody and Martin, 2009b) so that most studies are focused on the first few years after a wildfire limiting the acquisition of sufficient data to establish the statistical significance of observed physical relations. Second, the high-magnitude responses tend to be episodic and destructive with relatively short time scales of minutes to hours that make collection of field data difficult. Third, the response can depend upon the sequence of rainstorms of differing magnitudes (Germanoski et al., 2002; Moody et al., 2008a) making data interpretation complicated. Additionally, completely burned basins are often relatively small (hectares to a few tens of km² in size; Gartner et al., 2004), and nested within much larger regional basins. As a result, they can be affected by meso-scale ($\sim 1-10^4$ km²), short-duration and spatially variable precipitation. Such precipitation cells are frequently embedded within larger-scale regional precipitation patterns. To date, the characteristics of these storms have been far less predictable than those of regional larger-scale and longer-duration precipitation. Fourth, runoff and erosion from recently burned hillslopes do not always lead to significant floods or major erosion events such as debris flows, either due to inadequate rainfall or limiting geomorphic conditions (Cannon et al., 2001a, 2003; Larsen, 2003; Pausas et al., 2008; Robichaud et al., 2008a; Cannon et al., 2010). Runoff and erosion are unsteady (variable in time) and non-uniform (variable in space) processes, so that existing theories and methods developed for steady, uniform flows responding to steady, uniform precipitation must first be modified (Candela et al., 2005; Moody and Martin, 2009b). Fifth, the hydrological and geomorphic processes are typically non-linear and thus controlled by numerous physical thresholds (such as infiltration-excess overland flow and debris flow initiation), which add to the complexity. And lastly, investigations have been confounded by a multiplicity of measurement methods and scales, making comparisons between studies difficult (Shakesby and Doerr, 2006; Moody and Martin, 2009a).

A spectrum of post-wildfire hydrologic and sedimentologic responses ranging from no response to catastrophic floods, deadly debris flows, and damaging sedimentation has been documented in many different locations throughout the world (Rowe et al., 1954; Brown, 1972; Helvey, 1980; Scott and van Wyk, 1992; Shakesby et al., 1993; Cerdà, 1998a; Prosser and Williams, 1998; Conedera et al., 2003; Nishimune et al., 2003; Coelho et al., 2004; Lane et al., 2006; Tomkins et al., 2008; Silins et al., 2009; Cerdà and Robichaud, 2009; Dunkerley et al., 2009; Shin, 2010; Shakesby, 2011), and has provided a wealth of empirical data. A few of these field studies strive to understand and predict the underlying processes, but most have not (Larsen et al., 2009, p. 1394). The inability to accurately predict post-wildfire responses given all the different conditions such as rainfall characteristics, soil properties, and basin morphology (as summarized, for example by Shakesby and Doerr, 2006; Pausas et al., 2008; Moody and Martin, 2009a) highlights the need to organize these empirical data according to similarities to provide better understanding of post-wildfire processes. Better understanding of the processes will lead to improved predictive capabilities of post-wildfire models.

The purpose of this review is therefore twofold: first, to suggest an organizing framework to help synthesize the data by identifying common patterns and generalities that will help discover physical reasons for differences in responses, and second to identify priority research issues whose resolution will advance our understanding of the four major processes (precipitation, infiltration, runoff, and soil and sediment erosion and transport) controlling post-wildfire runoff and erosion responses.

2. Framework for organizing post-wildfire runoff and erosion responses

We propose an organizational framework that first groups the wide range of post-wildfire hydrologic and geomorphic responses into post-wildfire response domains. These domains are conceptually composed of a fire regime, precipitation regime, and hydro-geomorphic regime with broadly similar ranges of quantifiable metrics. The ultimate goal is to organize and synthesize the vast amount of empirical data from different post-wildfire domains in order to better understand each process, the reasons for differences in response, and to specifically predict, as close to real time as possible, the post-wildfire runoff and erosion response after new wildfires in any post-wildfire domain.

2.1. Fire regimes

Fire regimes are derived from fire behavior characteristics and effects (Agee, 1993; Neary et al., 2005). Characteristics include the temporal distributions (i.e. recurrence interval and fire duration), the spatial distribution (i.e. area affected and pattern), and the behavior (i.e. type of fire, combustion process, and vegetative type) (Neary et al., 2005; Krebs et al., 2010). Dominant drivers of fire behavior vary according to the different temporal and spatial scales (Whitlock et al., 2010a). Effects are most frequently assessed in relation to ecological (i.e. vegetation mortality and fire severity) and socio-economic (i.e. cost and damage to property) consequences (Krebs et al., 2010), but there has been little focus on the hydrologic and sedimentologic consequences. Most fire regimes have been described in a semi-quantitative way by using a quantitative metric of fire interval (alternatively fire frequency, fire recurrence interval, or fire rotation) but only a gualitative description for either burn intensity (Heinselman, 1981) or burn severity (Hardy et al., 1998; Brown, 2000; Keeley, 2009). For a given general area, fire intervals can be estimated using the 'individual-tree approach', 'composite approach' (Baker and Ehle, 2001), or a function of area (frequencyarea statistics, Malamud et al., 2005).

What is important for a post-wildfire response organizational framework is to identify those characteristics and effects of fire regimes that most directly affect hydrologic and sedimentologic responses and are readily available throughout the world. Fire intensity represents the energy released above ground and is not a good measure of the amount of heat transmitted into the soil (Neary et al., 2005). Burn severity generally has referred to ecological above-ground and below-ground effects (Neary et al., 2005). Burn severity does affect infiltration, runoff, and erosion, is patchy, and unfortunately is not readily available throughout the world. 'Depth of burn' is a metric that reflects the consumption of above-ground litter, duff, and woody material, which obstruct overland flow (Ryan, 2002); however, it is a qualitative metric. Soil temperature has been shown to have direct effects on chemical transformations and losses (Giovannini and Lucchesi, 1983; Giovannini et al., 1988), organic matter destruction, seed and plant mortality, water loss (Ryan, 2002), and soil erodibility (Moody et al., 2005). Soil temperature would be an ideal quantifiable metric; however, direct quantitative measurements of soil temperature are nearly impossible to make during wildfires in contrast to prescribed fires (Bento-Gonçalves et al., 2012), and thus, researchers have resorted to indirect qualitative metrics such as depth of burn and visual indicators of soil burn severity (Parsons et al., 2010). Post-wildfire hydrologic and geomorphic responses have been related to fire recurrence intervals (Swanson, 1981; Loomis et al., 2003; Shakesby and Doerr, 2006), and fire recurrence intervals are readily available so this metric seems to be the best quantifiable metric to characterize fire regimes for the purpose of an organizational framework. Estimates of fire intervals have been based on alluvial chronology (Meyer et al., 1995; Bigio et al., 2010), lake sediments (Roering and Gerber, 2005; Turner et al., 2008; Vannière et al., 2008; Whitlock et al., 2008), dendrochronology (Veblen et al., 2000; Heyerdahl and Alvarado, 2003; Kitzberger and Veblen, 2003; Swetnam and Baisan, 2003; Veblen et al., 2003), or frequency-area statistics (Nakagoshi et al., 1987 [Japan]; Archibald et al., 2005 [South Africa]; Díaz-Delgado et al., 2004 [Spain]; Malamud et al., 2005 [United States of America]; Jiang et al., 2009 [Canada]; Krawchuk and Moritz, 2009 [China]; O'Donnell et al., 2011 [Australia]; Berner et al., 2012 [Siberia], Kharuk et al., 2012 [Russia]; Sass et al., 2012a [Switzerland]; Tessler, 2012 [Israel]). Fire intervals based on frequency-area statistics are then given for a specified size or area of the wildfire, $A_F \ge 0.01 \text{ km}^2$ or \geq 10 km² (Malamud et al., 2005; Jiang et al., 2009). We refer the reader to several recent efforts to refine the fire-regime concept and to develop models to understand fire-climate interactions and socio-economic effects (e.g. Whitlock et al., 2010b; Bowman et al., 2011; Murphy et al., 2011). Fire intervals are affected by human population distribution and land use (Veblen et al., 2000; Pausas and Fernández-Muñoz, 2011; Pezzatti et al., 2011) and climate changes (Baker, 2003; Swetnam and Baisan, 2003; Roering and Gerber, 2005; Whitlock et al., 2010b; Pausas and Fernández-Muñoz, 2011), but the fire recurrence intervals probably will not change appreciably within the next decade, which is the intended timeframe of this review.

2.2. Precipitation regimes

Precipitation is the important driver of post-wildfire response. Quantifying precipitation regimes appears easier than fire regimes because of the numerous available metrics (see Section 3.2 for examples), yet deciding which metrics relate best to post-wildfire responses is more difficult. They include interval metrics that quantify temporally varying characteristics (such as total precipitation depth, intensity, duration, and recurrence intervals), sequence metrics that characterize the temporal sequencing of rainfall (Germanoski et al., 2002; Dunkerley, 2011), spatial metrics (such as extent and gradient of rainfall intensity), the type of precipitation (convective storm, cyclonic storm, hail, snow, and rain-on-snow), scale similarity (Menabde et al., 1997; Harris et al., 1998), and space-time properties (Eagleson and Shack, 1966; Venugopal et al., 1999; Bernardara et al., 2003). Depth-duration-frequency analysis characterizes rainfall regimes as the probable depth that would accumulate during rain storms with a duration of m-hours and a frequency of once per n-years (recurrence interval, Hershfield, 1961; Miller et al., 1973; Pilgrim, 1987; Institution of Engineers, 1998; Brown et al., 2010; NOAA, 2012). Because of the large spatial scale, river flood predictions often are based on rainfall durations of several hours (typically 6- and 24-h) and longer recurrence intervals (10-, 50-, or 100-year) in order to characterize the less frequent, rare extreme flood, which can be more damaging and thus are of more concern. Post-wildfire floods and debris flows can be as damaging, but generally respond to much shorter duration (5-, 15-, or 30-min) and more frequent (1-, 2-, or 5-year recurrence intervals) rainfall because of the changes in the rainfall-runoff processes caused by burning (Shakesby and Doerr, 2006; Moody and Martin, 2009b). Numerous papers have reported that peak discharge and sediment flux correlate with maximum rainfall intensities that are 30-min or less in duration (Moody and Martin, 2001a; Kunze and Stednick, 2006; Mayor et al., 2007; Spigel and Robichaud, 2007; Cannon et al., 2008; Moody et al., 2008a; Robichaud et al., 2008a; Cannon et al., 2010; Dunkerley, 2010; Cannon et al., 2011; Kean et al., 2011; Moody, 2012; Robichaud et al., 2013a,b), and that low-recurrence interval rainstorms can produce floods that normally are associated with a high recurrence interval.

These low-recurrence intervals can be as short as 2-years (Moody and Martin, 2001a; Conedera et al., 2003; Reneau and Kuyumjian, 2004; Kunze and Stednick, 2006). While analysis based on 1-hour duration rainfall is common, those for durations of less than 1 h are not; however, relations between longer and shorter durations have been developed (Hershfield, 1961; Miller et al., 1973). Thus, we suggest that the 2-year, 30-minute rainfall intensity be used as a quantifiable metric to characterize rainfall regimes associated with post-wildfire responses.

2.3. Hydro-geomorphic regime

The hydro-geomorphic regime can be quantified by such metrics as topographic slope, soil hydraulic properties, soil and sediment erodibility, and sediment supply. Slope essentially represents the force of gravity that drives the hydrologic (infiltration and overland flow) and sedimentologic responses (detachment, transport, and deposition). Slope here refers to a topographic facet (Daly et al., 1994), which has approximately the same orientation or aspect, rather than a point measurement. Slope alone is insufficient to completely characterize the regime in that the erosion component depends primarily on soil erodibility (Elliot et al., 1989; Flanagan and Nearing, 1995; Foster et al., 1995; Moody et al., 2005; Mataix-Solera et al., 2011) and secondly on the volume of stored sediment or sediment supply. We agree with Bryan (2000, p. 408) that "soil erodibility is not a single, simply identified property" but feel that the soil erodibility factor (K-factor) used by the Revised Universal Soil Loss Equation (RUSLE) is a good first approximation for an organizational framework (Renard et al., 1997). Values of the K factor $[L^{-3} T^3]$ are available for many regions throughout the world (Foster et al., 1981; van Rompaey et al., 2003; ASRIS, 2012), or can be predicted using the conventional soil erodibility nomograph (Wischmeier et al., 1971; Wischmeier and Smith, 1978), or alternatively, site-specific values can be calculated based on soil physical and chemical properties (Lu et al., 2001; Vaezi et al., 2011). Slope is easily computed for digital elevation models, but the appropriate horizontal scale needs to be selected to represent the process. These two variables can be combined into one by considering the usual form of the equation for soil detachment (Foster et al., 1977, 1995; Moody et al., 2005), D = kx, which assumes that the amount of eroded soil per unit area, D, is proportional to some driving force, rain energy, or flowing water, x (such as rain intensity, boundary shear stress or stream power) where the proportionality constant k is a form of the erodibility. Shear stress and stream power contain the slope, S, so that a possible hydro-geomorphic metric is the product of the soil erodibility *K*-factor and slope or *KS* $[L^{-3} T^3]$.

2.4. Post-wildfire response domains

Post-wildfire response domains can be quantified by the fire recurrence interval; the 2-year, 30-minute rainfall intensity; and the erodibility-slope product. The three metrics proposed here have a precedent in that they are similar to those used by Swanson (1981, p. 411) "to rank ecosystems in terms of fire's potential for impacting geomorphic processes ..." and to define a wildfire impact index based on physical principles of sediment transport (Moody and Martin, 2004). For example, in southern California the post-wildfire response frequently takes the form of sediment-laden floods and debris flows (Cannon et al., 2008, 2011; Kean et al., 2011; Schmidt et al., 2011; Robichaud et al., 2013b) and sediment-limited responses are rare. This domain has a Mediterranean-type climate, low elevation (maximum, 400 m), and fire-adapted chaparral vegetation with fire intervals <10 years (Keeley et al., 2008). The precipitation regime typically comprises long-duration winter frontal storms with some embedded cells of high intensity rainfall (Pacific-Medium rainfall regime; median 2-year, 30-minute intensity of 30 mm h^{-1} : Moody and Martin, 2009a) and the hydro-geomorphic regime is tectonically active with steep, rugged mountain ranges with median slopes of 0.56 (Moody and Martin, 2009a), resulting in an abundance of erodible material (median erodibility 0.034 t-ha⁻¹/MJ ha⁻¹ mm h⁻¹) and a median value of KS is 0.019 t-ha⁻¹/MJ ha⁻¹ mm h⁻¹. In contrast, the montane ecosystems often dominated by lodgepole pine (Pinus contorta; elevation 2500-2700 m; Weber, 1976) in the intermountain western United States have experienced infrequent wildfires (fire interval is 300 to 600 years; Romme and Knight, 1981; Westerling et al., 2011) whereas in the foothill ecosystems often dominated by ponderosa pine (Pinus ponderosa; elevation 1800-2500 m; Weber, 1976) have experienced more frequent wildfires (fire interval is 15-40 years; Veblen et al., 2000). Most runoff and erosion in both ecosystems are a response to high-intensity convective storms during the summer season (median 2-year, 30-minute intensity is 23 mm h^{-1} for the 1988 Yellowstone Fire in the montane ecosystem and 33 mm h^{-1} for the foothill ecosystem along the Colorado Front Range; Hershfield, 1961; Moody and Martin, 2009a; Moody, 2012). In the volcanic areas burned by the 1988 Yellowstone Fire median slopes are 0.40, erodibility 0.016 t-ha⁻¹/MJ ha⁻¹ mm h⁻¹, and KS is $0.0064 \text{ t-ha}^{-1}/\text{MJ} \text{ ha}^{-1} \text{ mm h}^{-1}$ (Moody and Martin, 2009a). In the granitic core of the Front-range Mountains of Colorado, the median slopes for multiple burned areas are 0.31, median soil erodibility is 0.014 t-ha⁻¹/MJ ha⁻¹ mm h⁻¹, and *KS* is 0.0043 t-ha⁻¹/MJ ha⁻¹ mm h⁻¹ (Moody and Martin, 2009a). These post-wildfire domains are less erodible than the southern California domains. Estimates of these quantifiable metrics can be plotted to identify domains with similar regime characteristics (Fig. 1) and maps showing the spatial variability of the three regimes for different wildfire prone areas in the world would be valuable as additional organizational aids.

Any specific post-wildfire response is composed of four major processes: precipitation, infiltration, runoff, and soil and sediment erosion and transport. These are linked by states (i.e. soil moisture and surface roughness) and by feedback mechanisms. For example, the land surface albedo changes after a wildfire (Wendt et al., 2007; Tryhorn et al., 2008; Montes-Helu et al., 2009; Beck et al., 2011). This may affect the precipitation process and precipitation changes soil moisture, which in turn affects runoff processes (Chen et al., 2001; Tryhorn et al., 2008). Prioritization of research issues in this review has been assessed based on the number of major processes that are dependent on the resolution of a research issue. Issues that involve feedback mechanisms (e. g. Sections 3.3, 3.4, and 6.1) increase their priority. Each process is discussed in a separate section of this review and can be read guasi-independently. Section 3 discusses precipitation, Section 4 covers infiltration and soil properties, Section 5 focuses on runoff, and Section 6 deals with soil and sediment erosion and transport processes. Future resolution of the priority issues in each section should provide specific quantitative physical relations for future modification or development of physically-based models of post-wildfire runoff and erosion responses.

3. Precipitation: research issues related to the temporal and spatial variability of meso-scale precipitation

3.1. Background

Precipitation is a primary variable for the other post-wildfire processes (infiltration, runoff, and soil and sediment erosion and transport) all of which have their own specific spatial and temporal scales. A priority issue is to determine the most appropriate parameterization to use in quantifying spatially and temporally variable rainfall as a variable for the other processes. These spatial and temporal scales may be interdependent because they are linked dynamically (Ormsbee, 1989; Foufoula-Georgiou and Krajewski, 1995; Venugopal et al., 1999; Bernardara et al., 2003), which may simplify quantifying and downscaling of combined space-time rainfall fields to the meso-scale (1–10⁴ km²). Constraints on predicting the meso-scale



Fig. 1. An example of a two dimensional plot showing selected post-wildfire response domains. This organizational framework identifies broad regime similarities, which can then be further examined to explain response differences. See text for details of the Chaparral, Yellowstone-montane, and Colorado-foothill sites. The fire interval for the Pacific Cascades is taken from Swanson (1981); for Chaparral from Keeley et al. (2008); for Yellowstone-montane from Westerling et al. (2011), for Colorado-foothills from Veblen et al. (2000); and for southern Arizona from Swetnam et al. (1992). Rainfall regime and hydro-geomorphic regime metrics were taken from Table 2 published by Moody and Martin (2009a) (*K*-factor in Table 2 must be multiplied by 0.040142 to convert to $(t-ha^{-1})/(MJ ha^{-1} mm h^{-1})$.

rainfall distribution (defined as a scale sufficiently small to be unaffected by the Earth's Coriolis force) have limited the success of predicting post-wildfire runoff and erosion. However, the scale of measurement (i.e. point scale for rain gages and areal for radar) may be different from that of the 'effective' parameters needed in models (Beven, 1996). Hydrologic models for rainfall-runoff predictions are large-scale (10,000–100,000 km²; Finnerty et al., 1997; Leavesley and Hay, 1998; Smith et al., 1999; Yates et al., 2000) and are often based on 1- to 6-h mean areal precipitation, which obscures any meso-scale variability embedded within regional weather systems. It is this meso-scale variability, however that drives the rapid urban (Cowpertwait et al., 2004) and post-wildfire (Moody and Martin, 2001a; Etheredge et al., 2004: Cannon et al., 2008; Kean et al., 2011) responses at hillslope and small basin scales (1-10 km²). Meso-scale variability can change seasonally within a specific precipitation-regime, but also can vary between precipitation regimes, ranging from low-intensity, long-duration spatially homogeneous to high-intensity, short-duration spatially heterogeneous rainfall. Some recent meteorological research (Chen et al., 2001; Vivoni et al., 2009; Moreno et al., 2012) has focused on meso-scale, physics-based rainfall models (Grell et al., 1994; Skamarock, 2004), which attempt to incorporate meteorological processes at hillslope or small basin scales. Collaboration between post-wildfire researchers and meso-scale meteorologists will be needed to advance the measurement, understanding, and prediction of temporal and spatial variations in rainfall over burned areas. While the temporal and spatial variations of rainfall may be linked at the meso-scale, we explore them separately in the following sections.

3.2. Temporal variations in precipitation

Rainfall accumulation during a storm is a discontinuous function of time-punctuated by intermittent short-duration, rain-free interludes followed by rapid accumulation associated with rain cells embedded in a storm. In view of this pattern, we consider here two properties: first, the rainfall totals for a given period (interval parameters), and second, the nature of the rainfall within a storm and during a sequence of storms (sequence parameters).

Different parameterizations of rainfall intensity and its thresholds have been used as basic variables to predict infiltration, runoff, and subsequent floods (Reaney et al., 2007). For simplicity, most predictions of post-wildfire infiltration have assumed constant rainfall intensities (Robichaud, 2000; Woods and Balfour, 2008) in modeling infiltration into the unsaturated zone (Green and Ampt, 1911; Horton, 1939; Mein and Larson, 1973; Smith and Parlange, 1978; Kutílek, 1980; Smith et al., 2002). A few methods predict only runoff volume (curve number, Mockus, 1972; Hawkins, 1973, 1993; Foltz et al., 2009) without considering rainfall intensity, but most methods predicting runoff rates (such as instantaneous unit hydrograph, Sherman, 1932; or the curve number method, Soil Conservation Service, 1972) use constant rainfall intensities compatible with the use of design storms for flood risk assessment. Now, however, with the increased use of rain gages with high temporal resolution (enabling near-continuous measurements) and improved numerical models capable of handling long time series, the most important issue becomes: 'What is the appropriate time-averaging interval for meso-scale rainfall that best quantifies rainfall properties as drivers of post-wildfire response' (Fig. 2)? For example, 10-minute rainfall intensities were found to correlate best with soil erosion rate (Spigel and Robichaud, 2007), 15-minute rainfall intensities were found to correlate best with debris-flow timing (Kean et al., 2011), and 30-minute rainfall intensities were found to correlate best with peak flood discharge (Moody, 2012). A problem with averaging data is that it may result in the loss of important process details such as lag times between peak rainfall and peak runoff (Yu et al., 1997). Rainfall is usually expressed as a depth accumulated during a given interval (e.g. 5-min, 1-h, or 1-day) or as the time taken to accumulate a constant amount (e.g. 1-mm or 0.01 inch). Rainfall metrics that have been used include:

(1) Total precipitation amount (Robichaud et al., 2006a, 2008a, 2008b; Cannon et al., 2010; Robichaud et al., 2013a, b);



Fig. 2. Temporal variability of meso-scale rainfall. Examples of cumulative rainfall profiles collected in an area burned by the 2010 Fourmile Canyon Fire in Colorado. The cumulative data are basic rainfall data from a 15-cm diameter tipping-bucket rain gage where each tip represents 0.254 mm of rain and have irregularly spaced time intervals. The elapsed time for 7 Sep 2011 has been divided by 10 in order to show it here on the same plot with the other rain storms. The problem is what sampling interval, Δt , is appropriate to use to relate the rainfall intensity to observed runoff and erosion processes. The corresponding rainfall profiles are shown in Fig. 3.

- (2) Average storm intensity (Cannon et al., 2010);
- (3) Intra-event rainfall intensities for varying duration (Moody and Martin, 2001a; Kunze and Stednick, 2006; Robichaud et al., 2006a; Mayor et al., 2007; Spigel and Robichaud, 2007; Cannon et al., 2008; Moody et al., 2008a; Robichaud et al., 2008a, b; Cannon et al., 2010; Dunkerley, 2010; Cannon et al., 2011; Kean et al., 2011; Robichaud et al., 2013a, b);
- (4) Al₁₅ (amount of rainfall multiplied by the maximum 15-minute intensity; Mayor et al., 2007)
- (5) A theorized geomorphically effective rainfall intensity (Flanagan and Nearing, 1995);
- (6) Rainfall erosivity (Renschler et al., 1999; Kunze and Stednick, 2006; Pietraszek, 2006; Wagenbrenner et al., 2006; Spigel and Robichaud, 2007); and
- (7) Temporally continuous intensity-duration relations (Cannon et al., 2008, 2011; Staley et al., 2012).

Although some of these interval metrics seem to make intuitive sense, only a few have been directly linked to physical processes in burned areas (Kean et al., 2011), and some have been used to define geomorphic thresholds (Cannon et al., 2011). Multivariate statistical analysis has been used to identify the most significant rainfall metrics to explain post-fire debris-flows and their volumes (Gartner et al., 2008; Cannon et al., 2010), but the physical significance of such metrics is unclear, so that future research should focus on establishing a physical basis for quantifying rainfall at time scales determined by the spatial scale of interest—i.e. point, plot, hillslope, or basin.

A basin's temporal runoff response to rainfall is inherently linked to spatial scales, determined by the infiltration process, the drainage network pattern, and flow velocities within this network. Rather than being a direct response to rainfall, runoff responds to the actual rainfall intensity minus the infiltration rate and topographic storage (effective rainfall). This produces a non-linear response (Overton, 1970) often characterized by thresholds (Doehring, 1968; Inbar et al., 1998; Reneau and Kuyumjian, 2004; Kunze and Stednick, 2006; Kean et al., 2011; Moody, 2012) and other factors that need to be considered in selecting a suitable scale for quantifying rainfall. For example, time-to-concentration (Barfield et al., 1981; Dick et al., 1997) may be an appropriate time scale to determine the time-averaging interval for rainfall, but in some precipitation regimes in some arid and semi-arid landscapes (which may respond similarly to bare, burned areas in terms of runoff and erosion) the time-to-concentration for continuous flow down slopes is often longer than the duration of the rainstorm (Yair and Raz-Yassif, 2004). With an increase to the basin scale, the runoff response to high-frequency (short-duration) fluctuations in rainfall intensity is damped (Eagleson and Shack, 1966) by the increase in and variability of times (time-to-concentration) required to route water through the different parts of the drainage network (Milly and Wetherald, 2002; Wainwright and Parsons, 2002; Smith et al., 2004). These network effects on basin response have been investigated using the geomorphic instantaneous unit hydrograph (Rodriguez-Iturbe and Valdes, 1979; Rinaldo et al., 1991), which initially assumed effective rainfall and constant drainage velocity although the latter assumption was later modified to include variable velocities (Robinson et al., 1995; Saco and Kumar, 2002). Additionally, as spatial scale increases, the threshold rainfall intensity for runoff generation in semi-arid landscapes increases from about 4 mm h^{-1} for an area of 10 m^2 to 22.5 mm h^{-1} for an area of $10^6 m^2$ (Cammeraat, 2004). Thus, the rainfall interval metric must take account of the following factors: spatial scale, geomorphic characteristics of drainage network patterns, infiltration process, the effects of the remaining patchy distribution of litter or duff, and the distribution of topographic depressions (see Section 4).

Parameterizing the rainfall intensity as the total rainfall over a given time interval gives no information about the temporal sequence of intra-storm rainfall intensities. The detailed sequence within a storm may be important in terms of process response (Beven, 1996)

and downscaling or disaggregation methods, assuming the distribution in shorter intervals is proportional to that in larger intervals (Ormsbee, 1989). Temporal sequences can have very different patterns but the same averages and peak intensities. Intensity may rise rapidly, reach a peak and gradually decline (Fig. 3A), or it may gradually rise towards the peak (Fig. 3C) or there may be several peaks with gaps of variable duration (Fig. 3D). Early efforts to characterize temporal rainfall variability focused on classification. For example, based on the result that the majority of rain measured in a network of 49 recording rain gages in east central Illinois, USA fell within a brief period of the total rainfall profile, a classification scheme was based on dividing the rainfall profile into four equal parts and determining in which quarter (1st, 2nd, 3rd, or 4th) the heaviest rain fell (Huff, 1967). This temporal rainfall variability is often referred to as the rainfall pattern (Smith, 1972; Xue and Gavin, 2008) or event profile (Dunkerley, 2011). It can affect infiltration response (Winchell et al., 1998; Xue and Gavin, 2008), time-to-ponding (Smith, 1972), runoff response (Reaney et al., 2007; Dunkerley, 2011), and erosion response (Lanini et al., 2009). Differences in the sequence of rainfall intensities within a rain storm may explain why the same basin can respond differently to rainstorms with similar interval metrics (Dick et al., 1997; Reaney et al., 2007). Also storm metrics such as coefficient of variation and the time between storm pulses have been shown to affect runoff coefficients and runoff flow distances (Reaney et al., 2007).

3.3. Spatial variations in precipitation

Wildfires are common in mountainous terrain with complex topography, and thus, spatial variations in rainfall are linked to topography and to variations in landscape surface properties. Regional orographic effects on rainfall associated with cyclonic storms have been known for a long time (Spreen, 1947; Burns, 1953; Linsley, 1958), and these effects have more recently been incorporated into digital models (e.g. Precipitation-elevation Regressions on Independent Slope Model, PRISM; Daly et al., 1994), but less is known about effects on smaller meso-scale, convective-storms. Spatial analyses of convective rainfall data indicate that there are "genesis zones" or "hotspots" with higher intensity rainfall than in surrounding areas (Henz, 1973; Banta and Schaaf, 1987; Williams and Moody, 2003; Hanshaw et al., 2008). They are common in the terrain of the Colorado Rocky Mountains and probably exist in other precipitation regimes. Although the scale of these hot spots has been identified for some precipitation regimes, the effect on runoff remains unknown. Future research needs, first, to identify them and understand the physical processes that create them. Further collaboration with meteorologists is needed to develop ways of quantifying rainfall in far more detail than is currently done using rain gage networks and conventional weather radar. Some possibilities are using portable C-band (Jorgensen et al., 2011) and X-band radars (Matrosov et al., 2013) that can collect dual-polarized measures of rainfall intensities at appropriate scales necessary for characterizing spatial and temporal properties of meso-scale storms as they pass across burned terrain. An example is the concept of a 'footprint' or the size of the storm, which may depend on the topography. Analysis of historical Doppler radar has shown that footprint size depends on rainfall intensity, which decreases exponentially (Fig. 4) with the increase in area coinciding with the 30-minute maximum intensity (Williams and Moody, 2003). Analysis of dual-polarized rainfall measurements over specific topography also can be used to improve rainfall downscaling methods (Ormsbee, 1989; Cowpertwait et al., 2004) for estimating temporal as well as spatial characteristics of meso-scale rainfall, and for identifying and understanding storm trajectories. Post-wildfire basin response depends upon the direction of storm movement relative to the drainage network in unburned (Ogden et al., 1995) and burned basins. High resolution rainfall data from these radars



Fig. 3. Examples of rainfall intensity profiles measured in an area burned by the 2010 Fourmile Canyon Fire in Colorado. These rainfall intensity profiles correspond to the cumulative profiles in Fig. 2. Profiles for 10, 13, and 19 July 2011 correspond to short-duration, high-intensity convective storms and the interpolated sampling interval was equal to 1-min. The profile for 7 September 2011 corresponds to a long-duration, low-intensity frontal storm.

could improve input into a stochastic daily weather generator such as CLIGEN (Nicks et al., 1995), which is now parameterized by the regional-scale PRISM (Daly et al., 1994) and the Rocky Mountain climate generator (Rock:Clime; Elliot et al., 1999; Robichaud et al., 2007b; Elliot and Hall, 2010) models. These models are used in the Water Erosion Prediction Project (WEPP; Flanagan and Nearing, 1995) as part of the Erosion Risk Management Tool (ERMiT; Robichaud et al., 2006b). However, the CLIGEN and Rock:Clime models now assume that storms have fixed time-varying intensity specific to each site, are uniformly distributed over the model's spatial domain, and are stationary-all of which limit their performance. In many burned basins, the spatial variability, or patchiness, of rainfall during a storm can be so great that the concept of a constant-intensity design storm is unrepresentative of actual conditions and therefore has limited use (Moody and Martin, 2001a) because antecedent conditions (e.g. soil water content and surface-storage detention) are often neglected (Ormsbee, 1989).

In addition to topography, the spatial distribution of landscape surface properties can influence meso-scale rainfall through feedback mechanisms. Generally, these properties include soil characteristics and micro-topographic effects on surface water detention. Conceptually, soil hydrology acts like a low-pass filter (Wu et al., 2002) of high frequency rainfall. Water is stored in the soil and released slowly back to the atmosphere. This provides a feedback mechanism, whereby rainfall from an earlier storm can directly affect later storm rainfall. Thus, the sequence of rain storms linked by soil moisture conditions can affect the post-wildfire response (Chen et al., 2001; Germanoski et al., 2002). Since burned areas have a lower albedo (Chen et al., 2001; Randerson et al., 2006; Tryhorn et al., 2008; Tsuyuzaki et al., 2009) and tend to dry more quickly than vegetated areas (Moody et al., 2007), surface temperature can also be higher than on the surrounding unburned areas. Such a warm surface temperature anomaly was created by an area burned by the 1996 Buffalo Creek Fire in the Colorado Front Range Mountains and cool surface anomalies were created by previous rain showers near the burned area (Chen et al., 2001). This temperature distribution had a sufficiently large effect to cause a convective storm to be confined to the warmer (~6 °C higher) burned surface, resulting in a 100-year rain storm over the burned area (Chen et al., 2001). The size of the burned area (4700 ha) may not by itself have been sufficient to create the local convective circulation, but instead may have triggered convective rainfall, which was poised near a threshold. Thus, the spatial distributions of soil properties provide a possibly important link between the precipitation regime and burned landscapes.

3.4. Future research priorities

Meso-scale rainfall is known to be the primary driver for post-wildfire runoff and erosion response, and is characterized by short-duration, high frequency temporal and spatial variability, which is influenced by complex topography and soil surface properties. How to quantify its effects will require collaborative research with meteorologists, which has so far been largely absent. The four research priorities below are related to the temporal and spatial variability effects of meso-scale rainfall on post-wildfire runoff and infiltration. The first two are broad and concern the role of soil properties in linking precipitation to infiltration and runoff. The third priority addresses a possible feedback mechanism between soil



Fig. 4. Areal footprint of rainfall versus rainfall intensity. Relation between one possible rainfall variable, the maximum 30-minute rainfall intensity (I_{30}), and the coincident rainfall area or 'footprint' on the ground. The data are from the analysis of historical Doppler radar data for 20 storms (1995–2001) over an area burned by the 2002 Hayman Fire in Colorado (Williams and Moody, 2003).

properties, and the fourth considers topographic effects on meso-scale rainfall.

- (1) Which time-interval metrics best explain post-wildfire infiltration and runoff responses at a given spatial scale, and how do these metrics change with spatial scales or post-wildfire domain? This will require better temporal resolution of rainfall and improved understanding of post-wildfire spatial variation in infiltration rates.
- (2) Which sequence metrics best represent the within-storm temporal change (rainfall profile), the pattern of multiple storms, and dry intervals, and how do they influence post-wildfire infiltration and runoff response? What factors control the key rainfall sequence metrics, and are there similarities across post-wildfire domains?
- (3) What is the magnitude of the possible feedback mechanisms in a burned area between the spatial distribution of soil properties and that of the meso-scale rainfall? What is the dominant control on this feedback mechanism (soil moisture, burn severity, or vegetation), and how does the feedback change over time with post-wildfire recovery?
- (4) How does topography affect the spatial distribution of meso-scale rainfall, and how important is it in controlling the locations of genesis zones or hotspots of persistent high intensity rainfall and the rainstorm 'footprint'?

4. Infiltration: quantifying soil properties and hydraulic effects

4.1. Background

Wildfires cause several geomorphically important changes in soil properties, including modification of the pre-fire soil profile and development of spatial variation of soil properties. Combustion removes some or all of the litter and duff layers (Robichaud and Miller, 1999), modifies the mineral soil layer at varying depths depending upon the fire regime (Doerr et al., 2009), may induce or enhance a water-repellent layer (Krammes and Osborn, 1969; DeBano, 2000; Doerr et al., 2000; Coelho et al., 2004; Woods et al., 2007; Finley and Glenn, 2010), and deposits a surface layer of ash (Fig. 5) containing varying quantities of mineral soil and organic substances as a result of increased wind transport during the wildfire (Byram and Martin, 1970). This ash layer is generally hydrophilic (Kinner and Moody, 2010; Woods and Balfour, 2010), though hydrophobic ash has been reported (Bodi et al., 2011). Increased runoff from burned areas can be caused either by infiltration-excess or saturation-excess overland flow or by some combination of both (Sheridan et al., 2007; Onda et al., 2008; Ebel et al., 2012). Increased runoff is often attributed to soil water repellency (SWR), the increase in amount of bare ground (Benavides-Solario and MacDonald, 2005; Larsen et al., 2009), the decrease in canopy interception (Stoof et al., 2012), and the lack of any surface water storage. Any remaining unburned duff 'layer' (partially decomposed litter with humus, Robichaud and Miller, 1999) below the ash layer can also create water repellent patches when dry, yet water absorbent patches when moist. This patchiness increases the spatial variability of the soil properties (Moody et al., 2007) and adds complexity to understanding post-wildfire runoff and erosion responses. Even when SWR is extreme, prolonged rainfall can cause the soil to be transformed to a 'normal' wettable state (Doerr et al., 2000; Stoof et al., 2011a), but soil can regain its repellent state once dry conditions return (Shakesby et al., 2000). Additionally, the effect of water repellency decreases with an increase in spatial scale (Larsen et al., 2009). Low runoff has been documented in areas where SWR is high (Stoof, C., Cornell University, per. commun. 2013), and high post-wildfire runoff has also been documented where SWR is absent, thus indicating that the presence of SWR is not always necessary for producing extreme floods (Meyer and Wells, 1997; Cannon et al., 2010).

The properties and spatial distribution of ash (Goforth et al., 2005; Kinner and Moody, 2008; Pereira et al., in press), duff (Robichaud and Miller, 1999) and fire-affected mineral layers (Ulery et al., 1996; Woods et al., 2007) are not well known and research has focused on the plot scale, sometimes the hillslope scale (Robichaud and Miller, 1999), but not on the basin scale. However, the limited results obtained indicate sufficient spatial variability to suggest that the concept of an ash or SWR 'layer' is perhaps misleading and should be replaced by one of 'patches' or more specifically a 'mosaic of patches with varying thickness' (Woods et al., 2007; Kinner and Moody, 2010; Pereira et al., in press). This concept of patches is supported



Fig. 5. Black ash layer about 1 cm thick and the underlying mineral soil layer in a burned area in Colorado.

by DeBano et al. (1998) who stated that "water repellency produced by fire is usually confined to areas beneath plant canopies" (Fig. 6). Thus, overland flow generated on these water repellent patches can infiltrate via wettable ash and soil patches (Shakesby et al., 2000; Doerr et al., 2009; Jackson and Roering, 2009). Soil heating below these patches can modify soil erodibility by destroying the organic



Fig. 6. (A) Burn severity map for the 2010 Fourmile Canyon Fire west of Boulder, Colorado, USA. This Burn Area Reflectance Classification map (BARC map) is based on grouping values of Δ NBR. The large-scale spatial variability in severity is evident. (B) Small-scale spatial variability of fire severity associated with the spatial distribution of vegetation on a hillslope burned by a wildfire in the Mojave Desert, California, USA. Areas of white ash indicate more complete combustion, and thus greater burn severity (caused by higher temperatures and/or longer duration) than areas of dark gray ash. Areas of light grayish-brown color are bare soil.

and chemical bonds (Giovannini and Lucchesi, 1983; Giovannini et al., 1988) within the soil to reduce the critical shear stress required to initiate erosion (Moody et al., 2005). Finally, severe wildfires can produce numerous burnt-out stump and root holes (Fig. 7), which augment the existing patchy pre-fire depression storage on the soil surface and may cause increased pipe flow. The magnitudes and spatial extent of these fire-affected patches are indicators of the burn severity.

4.2. Burn severity and soil hydraulic properties

Post-wildfire infiltration into the unsaturated zone is controlled by fire-induced changes in rainfall interception, soil-water storage, and soil hydraulic properties. Fire effects are often described by qualitative indices such as fire severity, burn severity (Keeley et al., 2008; Keeley, 2009), or soil burn severity (Parsons et al., 2010), which describe the above-ground organic matter consumption in qualitative terms (such as high, moderate, and low; Keeley et al., 2008), but do not directly relate to changes in soil hydraulic properties. Only recently, fire effects have been related to quantifiable metrics (bare soil exposed, fine root damage, water repellency, color change, and soil structure, Parsons et al., 2010). Organic matter consumption is spatially variable and often reflects the spatial distribution of pre-fire vegetation (DeBano et al., 1998) in addition to fire behavior (Fig. 6).

Fire effects on soil properties have been characterized indirectly by two burn severity metrics associated with two different spatial scales. The SWR metric (King, 1981; Letey et al., 2000) is measured on the ground at point to plot scales (1 cm²-100 m²) and is attractive because of its simplicity. Three SWR metrics are frequently used to assess burn severity: (1) the water drop penetration time test (WDPT, DeBano, 1981; Letey et al., 2000; Huffman et al., 2001), which represents persistence (Karunarathna et al., 2010) or the time needed for the contact angle to change to permit infiltration (Regalado and Ritter, 2009); (2) the molarity of ethanol droplet test (MED, Letey et al., 2000; Woods et al., 2007), which measures the critical surface tension; and (3) volume water that infiltrates in 1 min (1VOL) from a tension infiltrometer (Robichaud et al., 2008c). The other metric (change in the normalized burn ratio, ΔNBR , Key and Benson, 2004; van Wagtendonk et al., 2004) is measured remotely from satellite images at the hillslope to basin scales $(10^2 - 10^6 \text{ m}^2)$, and is attractive because of its large spatial coverage. The change in normalized burn ratio, ΔNBR , represents the difference between post- and pre-fire images of land surface reflectance for two bands measured by the Landsat satellite. One band is sensitive to green vegetation (near infrared) and the other to bare ground (short-wave infrared). The ratio ranges from about -100 (unburned) to 1000 (severely burned), and they are verified on the ground using standardized, but subjective, methods to assess soil burn severity (Parsons et al., 2010). Values of \triangle NBR are then grouped into qualitative values of high, moderate, and low soil burn severity classes, which are shown on burn severity maps (e.g. Chafer, 2008). Neither metric gives a direct measurement of fire-affected, soil-hydraulic properties such as hydraulic conductivity or sorptivity (Smith et al., 2002), but they do provide independent assessments of fire effects. Other metrics that might be promising include hyperspectral and multispectral remote sensing measurements (Kokaly et al., 2007; Robichaud et al., 2007c; Lewis et al., 2008).

Unfortunately, these burn severity metrics do not quantify the important soil-hydraulic functions required to study infiltration. Traditional infiltration theories require two soil-hydraulic functions (the soil–water retention curve, and the hydraulic conductivity function) to describe infiltration in the unsaturated zone (Smith et al., 2002). The soil–water retention curve is a relation between matric potential, ψ [L], and soil–water content, θ [L³ L⁻³], and the hydraulic conductivity function is the relation between θ and hydraulic conductivity, *K* [L T⁻¹], (Smith et al., 2002). Some work has been done to



Fig. 7. (A) Overland flow connectivity on hillslope burned by the 2010 Fourmile Canyon Fire west of Boulder, Colorado, USA. (B) Depression storage and root holes (shown as cross-hatched ovals in Fig. 10) also in the area burned by the 2010 Fourmile Canyon Fire west of Boulder, Colorado, USA. (C) Surface roughness, depression storage, and drainages burned by the Black Saturday 2009 wildfires in the Kinglake area of Victoria, Australia.

begin to quantify the effects of SWR (Ebel and Moody, 2012) and Δ NBR on the soil-hydraulic functions (Lewis et al., 2008) for fire-affected soils. Recent work (Bachmann et al., 2007; Karunarathna et al., 2010) for soils unaffected by fire has generated empirical models for soil-water repellency characteristic curves (i.e. SWR metrics versus θ and SWR metrics versus ψ). Thus, a research priority is to understand how SWR and Δ NBR are related to the soil-hydraulic properties of fire-affected soils. Achieving this will require more field and laboratory measurements of soil burn severity metrics together with measurements of the spatial variability of soil-hydraulic properties in burned areas (Berli et al., 2008). The results should establish a family of fire-affected soil-hydraulic functions for different values of each soil

burn severity metric, which can then provide the basis for modifying traditional infiltration theory.

4.3. Modification of traditional infiltration theories

Two components of traditional infiltration theories need to be addressed in order to modify these theories to predict post-wildfire infiltration. The first will be the development of the fire-affected soil-hydraulic functions described above, and second will be the inclusion of several special conditions related to wildfire. One condition is that post-wildfire responses are characteristically unsteady and transient so that the steady-state assumption is not applicable. Second is the assumption of the Green-Ampt infiltration model (Green and Ampt, 1911) used to simplify Richards equation (Richards, 1931) that the wetting front in a soil is sharply defined by uniformly wet soil above and 'dry' soil below (Hillel, 1998). This assumption is sometimes described as a 'moving piston wave' (Smith et al., 2002). Observations of the wetting front for fire-induced water-repellent soils (DeBano, 1981) and other water-repellent soils (Dekker and Ritsema, 2000) have indicated an irregular distribution of wet and 'dry' soil patches with depth, suggesting that the Green-Ampt model may not be appropriate for water-repellent soils. A third condition is that runoff from one water-repellent patch can infiltrate into another patch downslope (Sheridan et al., 2007; Larsen et al., 2009). This variability in the upper fire-affected 'layer' may be better characterized as a cumulative distribution function for the soil-hydraulic properties (Nachabe et al., 1997; Fiedler and Ramirez, 2000; Robichaud, 2000; Kinner and Moody, 2010). This approach has been incorporated into the Erosion Risk Management Tool (ERMiT) by using a distribution of hydraulic conductivity, which can also be spatially variable on a modeled hillslope (Robichaud et al., 2007a). Fourth, the increase in the amount of bare soil after a wildfire makes post-wildfire soils susceptible to soil sealing during subsequent rain storms by raindrop impact and runoff carrying ash and fine sediment particles (Martin and Moody, 2001; Larsen et al., 2009; Woods and Balfour, 2010). This fire-related condition can compact the upper soil layer with a corresponding increase in bulk density and a change in soil-hydraulic functions (Assouline, 2004). The extent of the sealing process needs to be measured, better understood, and incorporated in post-wildfire infiltration theories if appropriate. Fifth, recent observations of soil-water content immediately after a wildfire and before the first substantial post-wildfire rain storm have documented hyper-dry conditions with a threshold of $\theta < -0.02 \text{ cm}^3 \text{ cm}^{-3}$ or matric suction $> -3 \times 10^5 \text{ cm}$ (Moody and Ebel, 2012a). For these conditions, the only available process for wetting soils is the slow diffusion-adsorption process (Moody and Ebel, 2012a). To adequately characterize rewetting, modified infiltration theories need to consider rewetting as a two-stage process in relation to a threshold. Finally, local conditions such as animal or insect burrows, burnt-out tree roots, stump holes, and high rock fragment content may provide significant megapore pathways for infiltration through otherwise impermeable water repellent soil (e.g. Shakesby et al., 2007). These megapore pathways (Fig. 7B) need to be quantified, and if their effects are important, incorporated into post-wildfire infiltration theory.

The existence of multiple and spatially variable soil 'layers' needs to be included in post-wildfire infiltration models (Berli et al., 2008), and an immediate consequence is deciding how to determine the appropriate thickness of each 'layer' (Fig. 8). In the case of the uppermost layer, this requires knowledge of the depth of influence of the heat from a wildfire, which can penetrate a few centimeters into the soil (DeBano et al., 1976; Hungerford et al., 1996; DeBano et al., 1998; Robichaud and Hungerford, 2000; Stoof et al., 2011b) and change soil properties (Humphreys and Craig, 1981; Certini, 2005; Moody et al., 2005; Neary et al., 2005; Mataix-Solera et al., 2011). At present, these changes are not sufficiently well quantified for use in



Fig. 8. Idealized two-layer post-wildfire soil system with an ash layer overlying the soil. The overlying ash layer thickness is proportional to the quantity of burned material, and is shown increasing with the degree of burn severity and decreasing with time due to removal by wind and water erosion. The duff layer is depicted as being uniform in this diagram, though in reality it varies spatially in thickness. Ash and soil have specific values of hydraulic properties like sorptivity, *S*, and unsaturated hydraulic conductivity, *K* as well as specific soil–water retention curves. *H* represents an unknown depth of influence of the system on runoff.

models of post-wildfire infiltration and runoff processes, so collaboration with fire-effects modelers, who have developed models such as FOFEM (First Order Fire Effects Model, Campbell et al., 1995; Reinhardt, 2012) that predict the soil temperature profile during different fire behavior and fuel conditions (such as grass, canopy, and smoldering), might be highly productive.

4.4. Measurements of soil hydraulic properties

One indirect method of determining soil-hydraulic properties is to solve them as theoretical model parameters that best explain the observed data. These theoretical models of infiltration are derived from the non-linear, partial differential Richards equation (Richards, 1931; Philip, 1969) by making simplifying assumptions to produce an analytical form of the equation that can be readily solved. Soil hydraulic properties are not, therefore, independent measures, but rather they depend on the simplifying assumptions of the model. These models have generally been developed for relatively rock-free agricultural soils and may not apply to rocky, mountain soils common to many burned areas or to any soil subject to heat. The fundamental property in Richards equation is the hydraulic conductivity, K [L T], which controls the gravity component of infiltration over time scales of hours to days. One method of measuring the saturated hydraulic conductivity, K_s , was to adjust K_s in a hydrologic runoff model to fit observed hydrographs (Yates et al., 2000), and another was to solve for K_s indirectly using the Hydrus 1-D model (Šimùnek et al., 2008)

using actual intra-storm infiltration rates measured during natural rainfall (Moody and Ebel, 2012b). Philips (1969) derived an approximate solution for the cumulative infiltration depth, I(t) [L], as a power series expansion in time, t [T], which is frequently truncated to I(t) = $S t^{1/2} + K_s t$, where $S [L T^{1/2}]$ is called the sorptivity and can be considered as a second hydraulic property that controls the capillary component of infiltration over short-time scales of minutes to hours depending upon the soil texture (White and Sully, 1987; Smith et al., 1999). Therefore, the sorptivity of fire-affected soils may be initially more important than saturated hydraulic conductivity at the time scale of convective rainfall, which is typically short and only lasts 20-60 min but common in many post-wildfire response domains. Sorptivity is an attractive proxy for post-wildfire infiltration, but it is not a single value like K_s . Rather it is a function of the soil-water content. More measurements of the spatial distribution of K_s and S are needed as well as the temporal changes in these properties after wildfire. Cumulative distribution algorithms can provide a means to quantify the inherent spatial variability associated with these hillslope and surface conditions (Robichaud, 2000; Kinner and Moody, 2010). However, at present, the few existing measurements for fire-affected soil depend on the method of measurement and the theory used.

Several more direct methods have been used to measure soil hydraulic properties. Most measure the cumulative infiltration, I(t), and focus on measuring K_s , which can range over several orders of magnitude $(10^{-5} \text{ to } 10^{1} \text{ mm h}^{-1})$, Rawls et al., 1982). The disadvantage of most methods is that they use some type of artificial wetting of the soil under constant conditions (Moody and Ebel, 2012b) applied to a small area (~1 m²) in the field (Robichaud, 2000) or reconstituted samples in the laboratory (Fox et al., 2007; Novák et al., 2009). The most common methods are the constant-head or positive-pressure devices such as ring, drip, falling head (Nimmo et al., 2009), and disk permeameter or constant rainfall intensity simulators (Robichaud, 2000; Martin and Moody, 2001; Pierson et al., 2001; Assouline, 2004; Kinner and Moody, 2008, 2010). Positive pressure methods (12.7-mm head, Parks and Cundy, 1989; 10-mm head, Sheridan et al., 2007; 5-mm head, Nyman et al., 2011) essentially push water into the soil and may overwhelm the factors controlling post-wildfire infiltration–especially if K_s is near zero. Therefore, these methods tend to produce high estimates of K_s (Cerdà, 1996; Nyman et al., 2010; Ebel et al., 2012) that are probably inaccurate given the relatively low values of K_s for most fire-affected soils $(10^{0}-10^{2} \text{ mm h}^{-1})$; Imeson et al., 1992; Robichaud, 2000; Yates et al., 2000; Martin and Moody, 2001; Rulli et al., 2006; Robichaud et al., 2007b; Moody et al., 2009; Neary, 2011; Nyman et al., 2011; Ebel et al., 2012). Tension infiltrometers eliminate the positive pressure problem (Robichaud et al., 2008c; Moody et al., 2009), but frequently are difficult to interface with the coarse mountain soils characteristic of some post-wildfire response domains. Rainfall simulators often use a single, unrealistic rainfall intensity (Kinner and Moody, 2010), and the drop-size distributions, kinetic energies, or temporal variations are unrepresentative of natural rain (Renard, 1985). The few available values of sorptivity for fire-affected soils were measured using a tension infiltrometer (Mini Disk, Decagon, 2006), and range from 4.5 to 49 mm $h^{-0.5}$ (Moody et al., 2009; Ebel et al., 2012). For comparison, this range is slightly less than the range (21–73 mm $h^{-0.5}$) for unburned 'dry' sand to clay textured soils (Table 8.1; Smith et al., 2002), but similar to undisturbed field soils (10-36 mm h^{-0.5}, White and Sully, 1987). If we assume that soil-hydraulic functions have been developed for fire-affected soils and infiltration theory has been modified to account for the special conditions of fire-affected soils (both discussed above), then the next issue is: Which of the above methods is most appropriate for measuring the spatial and temporal variability of K_s and S in the field? Clearly, a standard method needs to be adopted so that measurements made in different post-wildfire response domains are comparable.

4.5. Time dependence issue

Some existing methods used to predict the timing and magnitude of runoff and erosion account for the spatial variability of burn severity conditions at the hillslope scale (Robichaud et al., 2007c), but not temporal changes during a given storm (Robichaud, 2000; Robichaud et al., 2007a). A water repellency index has been proposed (Pierson et al., 2001, 2008) to characterize reduced infiltration following wildfire, but any such index has only been considered to operate in a static way in a model (Robichaud et al., 2007a). Similarly, soil-hydraulic functions often represent equilibrium conditions found in a laboratory and yet dynamic conditions exist in the field, which could alter these relations (Wang et al., 1997; Hassanizadeh et al., 2002; Scheuermann, 2008). Increased understanding of short-term changes in soil-hydraulic properties caused by wetting and drying during individual rainstorms and by seasonal freezing and thawing of the soil (where winter temperatures are sufficiently low) will improve post-wildfire prediction models used by burned area assessment teams, land-managers, and emergency-responders. Additionally, at present, predictive models use effective rainfall, which is calculated as the rainfall rate (easy to measure but difficult to predict) minus the infiltration rate (difficult to measure and to predict). Thus, it is essential to understand the infiltration process in burned basins over appropriate time scales in order to understand the time-dependent effective rainfall. This new understanding can be used to further improve predictive models of post-wildfire runoff and erosion.

4.6. Future research directions

Characterizing wildfire effects on soil-hydraulic properties is pivotal to understanding infiltration, runoff, and erosion and in order to improve post-wildfire models. The priority research issues are:

- (1) What are the quantitative relations between burn severity metrics, such as \triangle NBR, WDPT, MED, and 1VOL, and soil-hydraulic properties such as K_s and S? Once these relations are established, researchers can define a set of soil-hydraulic functions for fire-affected soil to use as one component in modifying traditional infiltration theories.
- (2) What is the relative importance of fire-related condition (fire effects), such as spatial variability of the wetting front, diffusion-adsorption process, soil sealing, mega-pore preferential flow pathways, multiple soil and ash 'layers', and the penetration depth of heating?
- (3) How can the dynamic effects of changing soil-hydraulic properties be incorporated into post-wildfire infiltration models? This is needed to meet the forecasting requirements of the post-wildfire community.

5. Runoff: linking precipitation and basin morphology to post-wildfire response

5.1. Background

Post-wildfire hydrographs are difficult to predict because of insufficient data on soil properties (see Section 4) and the lack of rainfall–runoff data for burned basins, which are typically ungaged (Moody et al., 2008a). Post-wildfire hydrographs are also difficult to measure because streambed elevations often change rapidly in response to erosion and deposition, and post-wildfire flows can damage instrumentation. Therefore, much of the hydrological research literature has focused on predicting peak discharge of post-wildfire floods by using the paleoflood method (e.g. Jarrett and England, 2002), the curve number method (Hawkins and Greenberg, 1990; Cerrelli, 2005; Foltz et al., 2009), or direct measurements from burned basins (Moody and Martin, 2001a; Moody et al., 2008a; Foltz et al., 2009; Kean et al., 2011; Moody, 2012), but little has focused on predicting flood timing relative to rainfall (Elliot et al., 2010).

Most rainfall-runoff prediction methods have been developed for unburned basins, for which there is a large volume of literature (e.g. Hawkins, 1973; Interagency Advisory Committee on Water Data, 1981; NRCS, 1986; Beven, 2000; Feldman, 2000; Ries and Crouse, 2002). The focus in this review is on the rainfall-runoff process, yet snowmelt runoff from burned areas may impact water quality but is outside the scope of this review. In general, rainfall-runoff methods assume temporally and spatially uniform rainfall (which is usually not applicable to burned areas in mountainous terrain); and runoff contributions to channel flow from the entire drainage basin area (which is also questionable). Two frequently used methods are (1) regional regression equations and (2) the curve number method (Foltz et al., 2009). Both methods require some type of channel routing model (e.g. Peckham et al., 2004; Peckham, 2008; HEC-RAS, 2012) to predict the hydrograph. The first method requires knowing or assuming the increase in post-wildfire runoff in order to compute a fire-effects "modifier", which unfortunately is essentially what the method intends to predict. Both methods have difficulties accounting for fire-effects. Use of the curve number method for burned areas has produced conflicting results and demonstrates the need for further research according to Springer and Hawkins (2005).

Essential to any runoff model is a sub-model that is able to predict post-wildfire infiltration (see Section 4) and hillslope runoff contribution to channels. Depending on the post-wildfire response domain, hillslope-runoff-generating processes may switch between infiltration-excess and saturation-excess overland flow (Dunne, 1978; Wondzell and King, 2003; Keizer et al., 2005; Onda et al., 2008; Ebel et al., 2012). Intra-storm proportions of the two runoff-generating processes vary on steep hillslopes in southern California (Schmidt et al., 2011), and it is possible that they may change over intervals that are longer than individual storms. Runoff generation by infiltration excess has been found to be more sensitive to the uncertainty associated with precipitation than by saturation excess (Winchell et al., 1998). Understanding how these processes operate after wildfires is critical to runoff and flood predictions.

5.2. Factors affecting post-wildfire runoff

Explanations for increased post-wildfire runoff can be grouped into three hypotheses: (1) reduced infiltration caused by soil-water repellency (SWR), (2) increased flow velocities and connectivity as a result of the increase in percent bare-ground, and (3) reduced infiltration caused by soil-sealing and air entrapment. The significance of SWR in promoting runoff and erosion was first identified by Krammes and Osborn (1969). They attributed increased erosion to decreased infiltration and changes in soil-hydraulic properties. However, the impact of water repellency has not been isolated for large-scale field conditions (Shakesby et al., 2000), and no mathematical relations have been proposed that relate degree of SWR to runoff. The so-called bare-ground hypothesis seems to have originated with the paper by Cerdà (1998b, Figure 5), which linked post-wildfire runoff to percent vegetation cover, and later links were demonstrated between percent bare ground and runoff by using rainfall simulations (Johansen et al., 2001; Pierson et al., 2009), laboratory experiments (Pannkuk and Robichaud, 2003), and indirect evidence based on increases in sediment erosion (Benavides-Solario and MacDonald, 2005) and rill formation (Berg and Azuma, 2010) for natural rainfall. An increase in bare ground also results in an increase in the connectivity of water-repellent soil patches (Shakesby et al., 2000; Doerr and Moody, 2004; Cawson et al., 2010; Nyman et al., 2010). A possible threshold of 60-70% bare ground was found to be related to the connectivity of the bare patches (Johansen et al., 2001) and seemed to explain much of the post-wildfire erosion caused by increase runoff. The soil-sealing hypothesis originated with the paper by Rowe (1948). The role of ash in sealing was proposed by Mallik et al. (1984), and the combined hypothesis, which attributes sealing to precipitation (structural seals; Assouline, 2004) and runoff (depositional seals; Onda et al., 2008; Cerdà and Robichaud, 2009; Stoof et al., 2010; Woods and Balfour, 2010), was suggested by Martin and Moody (2001), and tested by Larsen et al. (2009). Laboratory and field experiments have shown that air entrapment (Jarrett and Fritton, 1978; Wang et al., 1997, 1998) can decrease infiltration rates (Krammes and DeBano, 1965; Suhr et al., 1984) and thus increase runoff.

It is possible that various combinations of these hypotheses are important across the spectrum of post-wildfire domains. Several runoff factors (critical thresholds, contributing area, depression storage, hydraulic roughness, and hillslope- and channel-flow connectivity) related to these hypotheses can possibly influence post-wildfire runoff in addition to the fire effects on soil-hydraulic properties (see Section 4) and characteristics of rainstorms (see Section 3 and Dunne, 1978). These runoff factors are considered in the following sub-sections.

5.2.1. Critical thresholds

Fire effects tend to reduce saturated hydraulic conductivity and sorptivity below pre-fire values, which can result in the appearance of critical rainfall-intensity thresholds controlling post-wildfire runoff. Additionally, surface ash and fire-affected soil layers can result in a critical-rainfall-depth threshold associated with either saturation-excess or infiltration-excess overland flow (Onda et al., 2008; Kirkby, 2011). Runoff thresholds may be related to the connectivity of flow paths, and the effects of subsurface connectivity have been explored using percolation theory (Lehmann et al., 2007). The post-wildfire response literature has established rainfall-intensity thresholds (Fig. 9), above which peak discharges from burned areas increase substantially (Doehring, 1968; Moody and Martin, 2001a; Reneau and Kuyumjian, 2004; Kunze and Stednick, 2006; Moody et al., 2008a; Moody, 2012), and rainfall intensity-duration threshold above which destructive debris flows of different magnitudes can be expected (Cannon et al., 2008, 2011; Staley et al., 2012). The sequence and timing of rainstorms can also affect post-wildfire response (Germanoski et al., 2002; Moody et al., 2008b: Kean et al., 2011), which may be related to scale-dependent thresholds (Cammeraat, 2004; also see Section 3) associated with the rain-free interludes. Researchers thus need to understand how these critical thresholds operate at different spatial scales.

5.2.2. Contributing area

During a rainstorm, the contributing area probably does not coincide with the topographic basin (Beven and Kirkby, 1979). Methods to predict peak discharge often assume contribution from the entire basin, just the burned area, or just the area burned at high or moderate severity. The actual contributing area may in fact not be static but change with time and this dynamic concept was the basis for the partial-area and the variable-source-area concept developed for unburned basins (Betson, 1964; Smith and Goodrich, 2000) and generally used to predict saturation-excess overland flow. Adopting some form of the partial-area concept might significantly improve the understanding of infiltration-excess runoff generation in burned areas with spatially variable patches having a range of fire-affected soil-hydraulic properties (Sheridan et al., 2007). For example, the rate of increase in the contributing area in semi-arid areas (which might be considered to be surrogates for burned areas because of the amount of 'bare' soil) tends to be high for short travel distances, but tends to be lower for longer travel distances because the probability of encountering patches with high infiltration capacity increases (Jackson and Roering, 2009; Kirkby, 2011). This spatial variability has been represented by several types of cumulative distribution functions (Hawkins, 1982; Kinner and Moody, 2010), which can provide a spatial average of an effective value of the soil-hydraulic properties. Runoff begins when rainfall intensity exceeds this effective spatial average. Additionally, some runoff prediction methods simulate



Fig. 9. Relation between peak discharge per unit burned area (Q_u) and the 30-minute maximum rainfall intensity, I_{30} , for paired rainfall and runoff data from 19 mountainous basins in different post-wildfire response domains in the western United States. The relation is based on data from basins ranging in size from 0.25 to 25 km². The data indicate a distinct I_{30} rainfall threshold, which is 7.6 mm h⁻¹ (based on results for the first year after wildfire). Solid blue circles represent maximum $I_{30} < 5$ mm h⁻¹ and correspond to perennial streams. Maximum $I_{30} < 5$ mm h⁻¹ values are shown as black squares and generally correspond to ephemeral channels. Red triangles represent basins having an average value of the change in normalized burn ratio, $\Delta NBR = 581 \pm 5\%$. Red lines correspond to the relation for $\Delta NBR = 100, 300, 500, 700, and 900$. See report by Moody (2012) for more details. (For interpretation of the references to color in this figure legend, the reader is referred to the web version of this article.)

the spatially variable micro-topography as well as the rainfallinfiltration for shallow overland flow (Fiedler and Ramirez, 2000). The Relative Surface Connection (RSC) function determines the percentage of the depression storage (Fig. 7, and see 5.2.3 below) contributing to the outlet (Antoine et al., 2011). Contributing areas, like source areas for channel initiation, probably also depend on the topographic slope (Montgomery and Dietrich, 1988; Dietrich et al., 1992), which can be highly variable in mountainous terrain. All variables controlling the size of the contributing area (Fig. 10) during a rainstorm are as yet unclear, but it is reasonable to assume that it will depend on: (1) rainfall intensities; (2) the corresponding rainstorm "footprint" (see Section 3); (3) topographic depressions and sinks representing scattered "holes" in the contributing area, (4) the area and degree of burn severity, and (5) the storm trajectory and storm profile (see Section 3).

5.2.3. Depression storage and surface roughness

Depression storage may increase after wildfires in some forested areas because of the creation of burnt-out stump and root holes (Figs. 7B and 10), and by tree root throw during or immediately after wildfire (Gallaway et al., 2009). Some of these holes might represent additional depression storage that could create an additional runoff threshold similar to the fill and spill hypothesis for subsurface flow (Tromp-van Meerveld and McDonnell, 2006). This situation would contribute to the runoff at different times (Antoine et al., 2011). On the other hand, some holes might represent distinct flow pathways for infiltration (see Section 4) and thus reduce potential runoff (e.g. Ferreira et al., 2005; Martin et al., 2008). These effects have yet to be adequately quantified and need to be addressed.

Surface roughness is a small-scale property but it is important because it controls runoff on hillslopes and in channels through the frictional resistance parameter. Surface roughness can change after wildfire by the consumption of vegetation, litter and duff, and surface obstacles (such as branches, logs, and plant stems), and by the deposition of an ash layer of variable thickness. Post-wildfire surface roughness may appear smoother (Fig. 7C) and better connected than for pre-fire conditions at the hillslope scale (~10–100 m), but overland flow is still controlled by the large relative roughness (ratio of the micro-topographic heights, *d* [L], to the flow depth or hydraulic radius, *R* [L], Smart et al., 2002) at the point-plot scale (\sim 0.1–1 m). The relation between the friction factor (Manning's *n* or Darcy-Weisbach friction factor, *f*) has been investigated in channels (Nikora et al., 2001) and on some unburned hillslopes (Lawrence, 1997) where f \sim (d/R)² for low-relative roughness changes to f \sim R/d for high-relative roughness on hillslopes (Lawrence, 1997). Surprisingly few studies (cf. Rulli and Rosso, 2005; Wagenbrenner et al., 2010) of the effects of wildfire on surface friction have been published. Recent advances in terrestrial and airborne LiDAR continue to improve digital elevation model (DEM) resolution, so that the importance of depression storage and changes in surface roughness may be better understood with future research (Staley et al., 2010).

5.2.4. Connectivity of overland-flow generating areas

Soil patches with different size and soil-hydraulic properties combined with scattered depressions, burnt-out stumps, and root holes highlight the need to consider spatial connectivity of overland flow. Several publications have begun to address this issue. The Relative Surface Connection function (Antoine et al., 2011) calculates the proportion of depression storage connected to the drainage outlet derived from the effective rainfall intensity (see Section 3), but omits detention storage (i.e. additional water flowing over the surface). Some researchers have expressed this concept as the connectivity length scale (analogous to the correlation length scale in turbulence, Batchelor, 1982) of spatial patterns of soil properties such as soil moisture, whose value in a pixel is either above (+1) or below (0)a selected threshold (Western et al., 2001). This length scale can be calculated for isotropic patterns or along a flow path to provide a 'bulk descriptor of spatial variability' but it is scale dependent (Western et al., 2001) and not originally designed for runoff (Mayor et al., 2008). Other researchers have quantified the connectivity of runoff source areas (pixels depicting bare soil) by defining a 'Flowlength' index (Mayor et al., 2008) as the average length of all potential flow paths based on a binary map of sources (bare pixels) and sinks (vegetative pixels or micro-depressions) and a single flow direction algorithm. 'Flowlength' correlates with total runoff and total sediment yield and these correlations increase with storm size. Although the original primary focus was on vegetation distribution,

this approach could include micro-topographic roughness (depending upon the pixel size of the DEM) caused by surviving post-wildfire plant mounds and probably burnt-out stump and root holes. Cawson et al. (2010) measured the effect of the length of unburned patches downslope from burned patches, and Moody et al. (2008a) defined the process-based hydraulic functional connectivity variable specifically for burned basins. This variable incorporates the magnitude of the burn severity through changes in the normalized burn ratio (Δ NBR see Section 4) and the spatial sequence of Δ NBR along flow paths with weighting proportional to the upstream contributing area.

Internal or external thresholds (Schumm, 1980; Davenport et al., 1998; Cammeraat, 2004) are important in establishing connectivity. They are often reflected as an abrupt change in runoff response highlighting the non-linear nature of this response. Future post-wildfire research needs to examine these connectivity variables, and possibly others, to determine how best to represent hydraulic properties of connectivity in order to predict post-wildfire runoff and erosion. Advances in process-based erosion modeling now allow multiple overland flow hillslope elements to capture some of the spatial variability and connectivity (Robichaud et al., 2007b), but there is room for improvement. Moreover, easily available DEMs tend to have too coarse a spatial resolution (generally 30-m pixels but sometimes 10-m pixels) to provide sufficient resolution of hillslope drainages and channels needed to apply some of these connectivity variables. Thus, an increase in DEM resolution is essential to help advance this issue of post-wildfire research. However, some hydrological models may require lower resolution than the input data, and up-scaling procedures (e.g. Milzow and Kinzelbach, 2010) would be needed to provide connectivity variables measured at plot or hillslope scale for use at the basin scale.

5.3. The significance of geomorphic properties

In general, most burned basins are ungaged before a wildfire, which highlights the need to determine the important geomorphic characteristics that might be used to predict the timing and magnitude of runoff immediately after a wildfire. These characteristics can be derived from DEMs and readily incorporated into basin response models. Typically, large wildfires in the US average no more than about 10³ km² in extent (NIFC, 2011) and in southern Europe they are typically much smaller (e.g. European Commission, 2012). At this scale the detailed morphology of a burned basin, its hillslopes, as well as the valley drainage networks (Fig. 7C) could all influence the timing, shape, and magnitude of the flood or debris-flow hydrographs (Kirkby, 1976; Sun et al., 1994; Dick et al., 1997; Moody and Kinner, 2006; Kean et al., 2011). Specifically, hillslope and channel drainage morphology can affect the geomorphic instantaneous unit hydrograph (GIUH; Rodriguez-Iturbe and Valdes, 1979) used in flood hydrograph predictions (D'Odorico and Rigon, 2003). At present, though, information incorporated into the GIUH theory is limited to knowing the effective rainfall (see Section 3) and it assumes the same constant runoff velocity in the hillslope drainage networks as in channels. This requires unrealistically uniform roughness or friction factors. As described in the previous section, additional insight is needed to parameterize spatial changes in surface roughness, which would lead to incorporating variable velocities into models, and to improved predictions of time-to-peak and the peak discharge. Another approach has been to statistically evaluate multivariate combinations of larger basin geomorphic characteristics (such as relief ratio, basin ruggedness, and basin gradients) as predictors of the probability of post-wildfire debris flows and debris flow volumes (Cannon et al., 2010).

5.4. Future research directions

Post-wildfire runoff occurs predominantly in ungaged basins and generally only empirical relations are available to predict peak flood discharges and little information is available for predicting when a flood will start. Modified infiltration theory (see Section 4) needs to be included in runoff predictions in order to improve estimates of the effective rainfall that is used in most rainfall–runoff models. The additional priority research issues addressing the runoff process are:

- (1) What are the effects of connectivity and surface roughness on post-wildfire runoff? How should they be parameterized and how are they linked with the spatial distribution of soil-hydraulic properties (Section 4), depression storage, and geomorphic properties of drainage networks at hillslope and possibly basin scales?
- (2) How does the contributing area change with rainfall characteristics (Fig. 10) for different post-wildfire response domains, and which runoff thresholds might be connected to the contributing area?
- (3) At what scales are the geomorphic characteristics of drainage patterns important in predicting post-wildfire runoff?
- (4) What are the underlying scale-dependent physical causes for rainfall thresholds that affect the timing and magnitude of different post-wildfire processes? These causes are required in order to allow generalization and transfer of results to other post-wildfire domains.

6. Soil and sediment erosion and transport

6.1. Background

Variability in post-wildfire erosion responses is caused by differences in the runoff and erosion and transport processes that operate in a given domain. For instance, there are some post-wildfire domains such as the relatively flat terrain of the Alaskan-spruce bog forest that may produce no response, whereas others such as the steep, tectonically active terrain of the California-chaparral forest may produce catastrophic debris flows. Distinctiveness in responses is not caused



Fig. 10. A conceptual diagram showing how the morphological pattern of drainage networks in channels and on hillslopes (not shown) might affect the timing and peak values of runoff hydrographs. Three sub-basins are shown with different morphologies. There is the question of which rainfall metric should be used (see Fig. 2). The runoff hydrographs relate to two different rainfall intensities, I_1 (dotted line) < I_2 (solid line), and the corresponding contributing areas are labeled.

just by differences in topographic slope, but are complex responses (Schumm, 1973) to the temporal and spatial variability of fire-affected soils (Section 4), rainfall (Section 3), infiltration (Section 4), and runoff (Section 5) processes. These processes are often characterized by thresholds that lead to nonlinearities in the erosion response. The thresholds can be external (Schumm, 1979; Cammeraat, 2004) such as that for infiltration-excess runoff and those for rainfall-intensity associated with peak discharge (see Section 5.2.1) or they can be internal (Schumm, 1979; Cammeraat, 2004) such as the relatively abrupt temperature threshold at about 220 °C, which controls the magnitude of the critical shear stress required to initiate erosion (Moody et al., 2005), consumption of fine roots that often hold soil and soil aggregates together (Mataix-Solera et al., 2011), soil gradation (i.e. amount of fines), and the critical gradient needed to initiate dry ravel (Roering and Gerber, 2005). Additional complexity is caused by the non-uniformity in the spatial distribution of sediment sources where sudden pulses of sediment or water can change the transport process at each tributary confluence downstream (Santi et al., 2008), and feedback processes where sediment transport on hillslopes can change the surface roughness, causing changes in runoff patterns and sediment transport (Imeson et al., 1992; Kirkby, 2011).

Key variables in post-wildfire erosion responses are runoff and sediment availability. Explanations for increased post-wildfire runoff are discussed above (Section 5.2). Sediment availability denotes the sediment supply (its quantity) and its associated erodibility (cohesive soils) or mobility (non-cohesive sediment). Soil erodibility depends on the specific erosion process, has been measured primarily for relatively homogenous agricultural soils, but is rarely constant (Bryan, 2000). In the context of fire-prone terrain with heterogeneous soils, erodibility can be particularly variable in response to changes caused by heating during wildfires and to the intra- and inter-storm changes in soil moisture conditions after wildfires. Soil depth, heat-induced changes in soil properties such as critical shear stress (Florsheim et al., 1991; Moody et al., 2005) and soil aggregate stability (Mataix-Solera et al., 2011), SWR (Doerr et al., 2000, 2009) can also lead to differences, as can root characteristics (Gyssel et al., 2005; Shakesby et al., 2007; Moody and Nyman, 2012) (Fig. 11). Mobility of non-cohesive sediment in rills, gullies, and channels depends on critical shear stress to initiate motion for a given particle diameter (Wiberg and Smith, 1987). Traditional theories for small relative roughness (ratio of the particle diameter to the flow depth) predict an increase in particle mobility as the channel slope increases, but there is evidence that the critical shear stress relation differs for large relative roughness (Lamb et al., 2008) and steeper slopes, typical of post-wildfire floods and debris flows.

In general, most currently available erosion prediction models applied to forest environments have evolved from models developed using plot studies on agricultural soils on low-angled slopes (Bryan, 2000). The applicability of these models to post-wildfire conditions, therefore, has distinct uncertainties (González-Bororino and Osterkamp, 2004) because slopes are often steep (Wagenbrenner et al., 2010), with heterogeneous soil-hydraulic properties (see Section 4) and surface conditions (see Section 5). The complexities of these post-wildfire erosion responses have resulted in the development of probabilistic models for prediction (Robichaud et al., 2007b), which require information on post-wildfire soil erodibility that is currently rarely available. Given this complexity, it is not surprising that it is difficult to extract simple cause and effect relations. Because of the many types of erosion and transport processes, a brief summary of each process is described below.

6.1.1. Dry ravel

This process depends on gravity, wind, and animal activity rather than water to detach and transport soil (Anderson et al., 1959; Krammes, 1965; Rice, 1982; Florsheim et al., 1991; Gabet, 2003a; Moody, 2010). Therefore, it is sensitive to the angle of repose or critical gradient (Roering and Gerber, 2005) and the amount of sediment supply on the hillslope. Some of the dry ravel sediment is trapped by



Fig. 11. (A) Measurements of soil detachment as a function of depth below the soil surface were made in a small tilting flume on soil cores with an ash layer. The soil cores were collected in the area burned by the 2010 Fourmile Canyon Fire. (B) Root length density plotted as a function of depth below the soil surface for the measurements shown in A (Moody and Nyman, 2012).

vegetation (Florsheim et al., 1991; Lamb et al., 2011), so that it becomes mobilized when the vegetation is burned. The process is thought by some to be the dominant source of post-wildfire hillslope sediment redistribution in southern California (Wohlgemuth and Hubbert, 2008; Lamb et al., 2011). Sediment mobilized in this way can reach the channel directly and can also act as a source of sediment for debris flows. Thus, dry ravel can increase sediment availability in the channels while reducing availability on the hillslopes. It has been modeled for burned areas as a non-linear function of hillslope gradient with a transport rate coefficient, k [L² T⁻¹], (Roering and Gerber, 2005; Rulli and Rosso, 2005). Where wildfires are frequent and post-wildfire erosion is high, the soil production rate would be expected to limit soil thickness and distribution and thus the magnitude of the dry-ravel process (Roering and Gerber, 2005), but other research suggests that increased soil production rates are unnecessary to explain substantial soil losses by fire-induced dry ravel following successive wildfires (Lamb et al., 2011). Dry ravel can mobilize large quantities of sediment and since it is independent of rainfall it has a unique position in the understanding of post-wildfire erosion response in the face of expected higher temperatures, increased dryness, and more wildfires in the future. With more wildfires and more people living in fire-prone areas where this process is common, the need to understand dry ravel is likely to increase in importance.

6.1.2. Raindrop, rain-flow, and interrill erosion

Raindrop impact is one of the most obvious and effective detachment processes (Gabet and Dunne, 2003). It is important on burned hillslopes where initially litter and duff has burned exposing patchy areas of bare soil, and later when the patchy ash layer has been removed re-exposing bare soil. Raindrop impact can combine with shallow overland flow near the top of hillslopes to create rain-flow transport (Moss, 1988; Moody, 2010) that can change to interrill or sheet flow further downslope (Inbar et al., 1998; Moody and Martin, 2001c; Benavides-Solario and MacDonald, 2005; Rulli and Rosso, 2005; Mayor et al., 2007; Sheridan et al., 2007). The process depends on slope angle and has been modeled at the burned basins scale using the raindrop erodibility coefficient $\kappa_c \, [M^{-1} \, T^2 \, L^{-2}]$ (Rulli and Rosso, 2005). The detachment component of this process may also depend on air entrapment (see Section 5.2). This has been observed in laboratory flumes when trapped air escapes explosively carrying soil particles into the flow (Suhr et al., 1984; Moody et al., 2005; Moody and Nyman, 2012). However, this process has not been observed and measured in the field so that its magnitude and importance are unknown.

The combined effect of raindrop, interrill, and rill erosion can be estimated from the Universal Soil Loss Equation using the bulk soil erodibility *K*-factor $[L^{-3} T^3]$ (see Section 2.3), based on total sediment yield per unit rainfall erosive index from standard plots (Wischmeier and Smith, 1978). Interrill erodibility, alone, has been modeled by assuming erosion is proportional to the rainfall intensity with erodibility equal to K_i [M T L⁻⁴], (Foster et al., 1995). Why interrill erosion or sheetwash dominates in one post-wildfire domain (Fig. 12A) and rill erosion in another (Fig. 12B) may depend on fire-induced changes in soil erodibility and may be affected by root properties (Sheridan et al., 2007; Moody and Nyman, 2012).

6.1.3. Rill erosion

On some post-wildfire hillslopes, interrill or sheetflow erosion dominates (Sheridan et al., 2007) whereas on others, rill erosion dominates (Moody and Martin, 2001c; Robichaud et al., 2010; Wagenbrenner et al., 2010; Kean et al., 2011). Rill erodibility has been modeled by assuming erosion depends on either the excess shear stress (Hairsine and Rose, 1992; Foster et al., 1995; Wagenbrenner et al., 2010) with the erodibility given by K_r [T L⁻¹], or on the stream power (Wagenbrenner et al., 2010) with erodibility given by K_{Ω} [M L⁻¹]. Rill erodibility for areas burned at high severity has been found to be about three orders of magnitude greater than for unburned soils, and to decrease with time in response to changes in the available sediment over timescales of minutes (Pierson et al., 2008; Wagenbrenner et al., 2010). Rill erosion in steep terrain, where shear stresses and transport capacities increase rapidly with slope, may differ from that used in models developed for agricultural soils (Foster and Meyer, 1972) where slopes are less. Thus, future research into dynamic rill erosion for burned areas is required to better understand how: (1) erodibility changes with time; (2) steeper slopes can be accommodated; (3) changing sediment availability alters rill or channel shapes, and (4) changing rill or channel roughness affects rill erosion processes.

6.1.4. Landslides and slope failures

Landslides and slope failures depend on the shear strength of the soil. Shear strength is a function of soil pore-water pressure, internal friction angle, and root cohesion (Schmidt et al., 2001). Dry soils generally have negative pore-water pressure, which tend to augment shear strength (Rahardjo et al., 2005) and produce stable slopes, whereas rainfall infiltration or snowmelt can produce wet soils near saturation with low shear strength that is conducive to landslides or slope failures (Gabet and Mudd, 2006; Germer and Braun, 2011). Roots generally increase cohesion, shear strength, and slope stability. However, wildfires can kill trees and later as their roots decay, slope stability decreases and landslides and slope failures are more common (Benda and Dunne, 1997; Jackson and Roering, 2009). While old roots decay, new roots are growing so that root strength reaches a minimum 8–12 years after the trees have died, which leads to an





Fig. 12. (A) Sheetwash erosion with a braided pattern dominates on a hillslope burned by the Black Saturday 2009 wildfires in the Kinglake area of Victoria, Australia. Ash has accumulated in the hollow in the middle of the photograph taken in March 2009 about one month after the wildfires. (B) Rill erosion on a hillslope burned by the 1996 Buffalo Creek wildfire in the Colorado Front Range southwest of Denver, Colorado, USA.

associated increase in landslides and slope failures (Schmidt et al., 2001; Jackson and Roering, 2009).

6.1.5. Drainage, channel, and debris flow erosion

Drainages are hillslope depressions (Fig. 7C) where runoff is concentrated, but where the flow is insufficient to create a channel with distinct banks. Increased runoff from burned areas, however, can cause these features to become incised and form channels (Collins and Ketcham, 2005; Moody and Kinner, 2006; Moody et al., 2008b). At this scale, flow depth is often greater than roughness heights (low relative roughness) so that traditional sediment transport models for bed-load (Gomez, 1991) and suspended-load (Yang, 2006) may be applicable if sediment concentrations are low. But, an important caveat for the use of these models is that these channels are steeper than those for which traditional transport models have been developed, and changes in the frictional resistance equations are necessary (Armanini and de Silvio, 1991; Wohl, 2010; Lamb et al., 2011). Debris flows represent a special case of sediment-laden flow in channels. They are classed as non-Newtonian fluids with unconsolidated sediment concentration of 50-77% by volume creating an interstitial fluid of water and fine sediment resembling 'wet concrete' and capable of supporting gravel and boulders while flowing (Pierson and Costa, 1987; Costa, 1988; Iverson, 1997). They operate, however, for relatively short periods (<10⁴ s) (Iverson, 1997), and have been reported in many different post-wildfire response domains (Conedera et al., 2003; Van Dine et al., 2005; Cannon et al., 2010; Nyman et al., 2011; García-Ruiz et al., 2012; Sass et al., 2012b).

An important though poorly understood threshold is the triggering mechanism for debris flows in different post-wildfire response domains. One mechanism has been described as the progressive entrainment of soil eroded from hillslopes and channels by overland flow (Wells, 1987; Meyer and Wells, 1997; Cannon et al., 2001a, b; 2003; Cannon and Gartner, 2005; Santi et al., 2008) coupled with the role of ash to provide sufficient fine-grained material (Gabet and Sternberg, 2008) to support the sediment. The change from sediment being supported by fluid-turbulence forces to sediment being supported by fluid-turbulent and solid forces (Iverson, 1997) and the sediment entrainment process (McCoy et al., 2012) are not understood. A second possible mechanism is saturation of soil above the fire-induced water repellent 'layer', which initiates 'thin debris flows' (Gabet, 2003b). A third possible mechanism results from shallow landslides induced by infiltration into soils with low soil hydraulic conductivity, which increases pore pressure resulting in liquefaction and mobilization (Gabet, 2003b). This is a mechanism observed in unburned settings (Gabet and Mudd, 2006; Gabet and Sternberg, 2008) where the ratio of fines to sand-size particles appears to be as important as the hydraulic conductivity, but it cannot account for increases in the size of a debris flow (Santi et al., 2008). The first mechanism depends on soil erodibility, the second on the intensity and spatial variation in SWR patches, and the last on shear strength and hydraulic properties of subsurface soils.

6.2. Sediment availability

Wildfire directly increases sediment availability in two ways. First, the canopy, litter, and duff layers are burned, which increases the sediment supply by exposing large areas of bare soil on hillslopes to erosive forces once the protective layer of ash is removed from the system, and second, the heat pulse into the soil changes the erodibility by altering aggregate stability (Mataix-Solera et al., 2011), reducing the critical shear stress needed to initiate motion (Moody et al., 2005; Wagenbrenner et al., 2010), increasing transport rate coefficient (Roering and Gerber, 2005), and decreasing root cohesion (Moody and Nyman, 2012). However, the thickness of the erodible soil is unknown, and transient, as it is depleted by erosive forces. This eventually limits the sediment delivery to channels (Meyer and Wells, 1997; Desilets et al., 2007) as a source of sediment for floods and debris flows. Another source of potential sediment are stream banks in the riparian zone that may have been unaffected by wildfire. These sources represent an indirect increase in sediment availability as a consequence of the increased runoff response after wildfires. However, at present, no reliable methods exist for assessing the spatial distribution of these sources of sediment.

Some investigators have used time-since-last fire (Rowe et al., 1954) as a proxy for estimating the sediment supply, whereas others have used a more direct surveying method (Staley et al., 2010; Schmidt et al., 2011). This consists of differencing pre- and post-flood, high-resolution (millimeter to centimeter scale) terrestrial LiDAR surveys combined with detailed process mapping and characterization of geomorphic form to identify the locations of sediment sources (Fig. 13). Vegetation can trap sediment (Lamb et al., 2011), so that vegetation maps might provide estimates of the sediment supply for the dry ravel process (Gabet, 2003a). Determining the sediment availability for channel erosion will depend on understanding the sediment entrainment processes for flow in steep channels (Armanini and de Silvio, 1991; Takahashi and Sawada, 1994; Wohl, 2010; Lamb et al., 2008). However, additional research is needed into other fluvial and debris flow transport processes and exploration of remote sensing methods to determine to what extent the sediment supply in a basin might depend on basin and channel geomorphic and geologic properties (e.g. curvature, slope, drainage density, and bedrock geology). Determining the initial sediment availability and the changes in erodibility and supply is one of the most challenging issues of post-wildfire erosion prediction.

6.3. Post-wildfire erosion and transport measurements

In response to the variety of post-wildfire erosion and transport processes operating at different scales, a range of measurement methods have been used. Results measured at one scale cannot be scaled up or down unless the dominant process is known to have the same or similar temporal and spatial scales. These different scales are also part of the reason for the numerous methods developed to measure post-fire sediment response. Some examples are: (1) ground-level changes measured with erosion pins or a microprofiling device such as an erosion bridge; (2) surveyed hillslope or channel cross sections; (3) suspended-sediment samplers; (4) bounded and unbounded hillslope plots; (5) amounts of sediment trapped behind silt fences and debris dams; and (6) sediment accumulated in small reservoirs. Each has produced a different unit of measurement (Shakesby and Doerr, 2006; Moody and Martin, 2009a). The use of sediment yield (mass/horizontal area/year) has been inherited from past sedimentary research focusing on long-term, larger-scale questions comparing continental scale erosion over decades. The use of vield has continued, probably for the sake of comparisons, but it may not best represent small-scale erosion processes in burned basins, their episodic character (changing over minutes or hours), or the nature of post-wildfire erosion itself. Thus, a measure of sediment yield might be suitable for hillslope erosion processes, but other means of expression might be more suitable for channel erosion processes (mass/cross-sectional area/unit of time), impacts on fish habitat (mass/width/time), and denudation (depth removed/unit area/unit of time).

One problem with most field measurements of erosion is that they tend to be continued only over short periods (typically no more than 3 years) and generally span only part of the recovery period after a wildfire. Thus, long-term perspectives of wildfire impact are few (Heede et al., 1988; Cerdà and Doerr, 2005; Cerdà and Lasanta, 2005), and more are needed to understand the effect on ecosystems over longer timescales. Other methods that can provide both a perspective on the order of decades in length and a relatively large-scale view include the measurement of ¹³⁷Cs (Menéndez-Duarte et al., 2009), which has been used in conjunction with other cosmogenic radionuclides (Blake et al., 2009; Smith et al., 2011b,c), and assessment of the sediment volumes in reservoirs draining burned areas. The use of sediment tracers is a large field, and we refer the interested reader interested to the review by Smith et al. (2012). However, each method has its own sets of problems. Some often require very exacting conditions to be used successfully, and fire-related erosion may be difficult to separate from that produced between wildfires. Because the number of cosmogenic radionuclide applications relating to wildfire is relatively small (Smith et al., 2012) to date, there has been little opportunity to cross-check estimated erosion rates with those gathered using more conventional methods.

Prediction of erosion and deposition can be affected by the temporal variability of rain (Benavides-Solario and MacDonald, 2005; Rulli et al., 2006; Mayor et al., 2007) so that it is important to normalize erosion values according to some rainfall measure (e.g. rainfall total, rainfall intensity, or some function of rainfall intensity). Normalizing on a storm basis assumes all storms have the same duration. It also tends to bias results by excluding the long intervals between storms when there is virtually no erosion. Some published values of sediment yield have been normalized by total cumulative rainfall (Johansen et al., 2001) and rainfall erosivity (Spigel and Robichaud, 2007). All these different measurement units have made it difficult to compare post-wildfire sediment responses (Shakesby and Doerr, 2006; Moody and Martin, 2009a). An important objective for the



Fig. 13. A. Estimates of available sediment made by differencing two tripod mounted LiDAR surveys (September 2008 and December 2008) in a small watershed burned by the 2008 Gap Fire in California. Sediment source areas are shown in yellow-red-brown colors and depositional areas are shown in blue-violet colors (Staley et al., 2010).

future is to determine which standard measurement methods are appropriate for post-wildfire sediment research and how they can be implemented to ensure that vital, though not necessarily difficult, measurements are made (provided a researcher is aware of the need) in the future to provide comparable data. A means of speeding up the dissemination process might be to set up a website similar to that available for sediment charcoal data (International Multiproxy Paleofire Database Charcoal Sediment Data; http://www.ncdc.noaa.gov/paleo/impd/impd_char_submit.html).

6.4. Causes of differences in processes between post-wildfire domains

Post-wildfire erosion and transport are especially complex non-linear processes because they depend on a 'web-like' pattern of other non-linear processes such as meso-scale rainfall, infiltration into water-repellent soils, scale-dependent runoff, and sediment availability. It appears to be difficult to disentangle this web to identify why one process dominates in one post-wildfire domain and another dominates elsewhere. Perhaps, it should be assumed that all erosion and sediment transport processes are possible, but one or two are selected to dominate at any given time in response to the particular characteristics of the fire, precipitation, and hydro-geomorphic regimes. For example, in some domains, in response to specific storm conditions, channel erosion contributes more sediment than hillslope erosion (Moody and Martin, 2001b, 2009a; Santi et al., 2008), whereas in other domains (though also sometimes in the same domain) in response to less intense rainfall, most material is derived from hillslopes (Staley et al., 2010). In yet another domain, dry ravel is produced (without any rainfall) in abundance and contributions from hillslope and channels are considered to be about equal (Wohlgemuth per, commun, 1999; Wohlgemuth and Hubbert, 2008).

Surface erosion by infiltration-excess overland flow has been documented in many post-wildfire response domains (Martin and Moody, 2001; Kinner and Moody, 2010; Robichaud, 2000; Pierson et al., 2001, 2007; Nyman et al., 2011; Ebel and Moody, 2012; Lane et al., 2012), but does not appear to be important in others (Wondzell and King, 2003). The proportion of infiltration-excess and saturation-excess overland flow varies over long periods and shorter periods within individual storms (Schmidt et al., 2011; Ebel and Moody, 2012), and it is likely that the proportion of these mechanisms varies between post-wildfire response domains. Similarly, debris flows have different dynamics at the same site for different rainstorms (Kean et al., 2011), and debris flow initiation processes have been found to vary between runoff-dominated and infiltration-dominated (Cannon et al., 2001a, b; Jackson and Roering, 2009).

A proposed first step in understanding the reasons for the dominance of specific erosion and sediment transport processes in certain post-wildfire response domains would be to initially identify and compile in a database the quantitative metrics for the fire, precipitation, and hydro-geomorphic regimes (see Section 2) associated with each process, its magnitude, and any related thresholds. This would provide the initial framework. A second step could be to include measurements of the changes in the relevant soil and soil-hydraulic properties, which would allow further sub-classification of the processes within the organizational framework of the post-wildfire domains. These domain-specific datasets could then be synthesized and common processes, patterns, and generalities identified.

Measurements of soil properties and actual measurements of the magnitude of the post-wildfire erosion and sediment transport responses would need to be made using standard procedures in order to produce comparable data. Such a *modus operandi* would probably be best achieved through collaboration and consultation amongst experts. An international meeting or special sessions at meetings of interested scientists would provide the forum for the necessary debate leading to the choice of standard methods of measurements in order to improve comparability of results from around the world. Additionally, multiple sites could be identified and measurements continued for several years after a major wildfire (as a long-term program) to determine the impacts of wildfire on the environment. This would provide process-based data (collected using standard methods) for improving long-term post-wildfire erosion and transport models used by land and emergency managers.

6.5. Future research directions

Multiple post-wildfire processes can erode and transport soil and sediment, which depend on sediment availability. Each of the multiple processes summarized in Section 6.1 defines a different erodibility 'constant' linked to soil properties. However, erodibility is not a true constant but rather dependent on factors with different temporal scales such as soil-water content, SWR, organic matter, and sediment supply, but also on the spatial distribution of the fire-affected soil properties. The primary research issues are:

- (1) What are the relations between quantitative metrics for burn severity and erodibility parameters as a function of soil depth, soil types, and root properties?
- (2) What is the sediment entrainment and transport processes for flows in steep, rough channels?
- (3) What standard measurement methods can be used to assess the sediment supply on hillslopes and in channels?

7. Summary

A large body of empirical data and related physical understanding now exists concerning complex post-wildfire runoff and erosion processes for many different post-wildfire domains throughout the world. A common theme within each of the four major processes (precipitation, infiltration, runoff, and soil and sediment erosion and transport) discussed has been the need to understand their temporal and spatial distributions (Fig. 14). It is evident that post-wildfire responses are the result of the superposition of the spatial distribution of precipitation upon the spatial distribution of fire-affected soil properties and complicated by changes with time on different time scales. Soil properties have been shown to be a critical link between all major processes (Fig. 14).

Thus, the highest priority for future post-wildfire runoff and erosion research is to understand the relations between soil properties



Fig. 14. Schematic representation of the research issues for the four major processes of post-wildfire runoff and erosion response discussed in the text. The thin black lines divide the diagram into the four processes (precipitation, infiltration, runoff, and erosion), where the central yellow circle represents the factors associated with each process. Where a factor spans one or more of the processes (for example, 'soil burn severity'), then a link between the respective processes is indicated. The main research issues for each process are given in the outer blue circle. The ultimate goal is to organize and synthesize the vast amount of empirical data from different post-wildfire domains in order to better understand each process, the reasons for differences in response, and to specifically predict, as close to real time as possible, the post-wildfire runoff, erosion, and sediment transport response if a wildfire should burn an unburned basin in one of the post-wildfire domains. (For interpretation of the references to color in this figure legend, the reader is referred to the web version of this article.)

and burn severity metrics (which include soil–water repellency metrics). Soil properties include (1) the soil–water content that may affect meso-scale rainfall, (2) the soil-hydraulic properties (soil– water retention characteristics, sorptivity, and saturated and unsaturated hydraulic conductivity) that control infiltration, connectivity, and contributing area, (3) the physical changes in soil properties that cause sealing, depression storage and changes in surface roughness, and (4) the multiple soil erodibility parameters. All these soil properties may change abruptly during a wildfire and then change more slowly after it.

The second priority is to appropriately characterize meso-scale rainfall because it is the primary driver for post-wildfire responses. The physical basis for determining the time-interval metrics that best predict runoff and erosion and how these depend on scale need to be understood if what is learned in one post-wildfire response domain is to be applied to other less studied domains. Additionally, suitable sequence metrics must be selected to represent the effects of the temporal distribution of rainfall on post-wildfire runoff and erosion (Fig. 14).

The third priority is to develop methods to determine sediment supply and to modify existing sediment transport algorithms so that they can be used to predict entrainment of soil and sediment and transport through steep, rough channels. To help overcome the lack of sufficient runoff and erosion and transport data from burned basins, the importance of morphological characteristics of the burned basins needs to be investigated so that their effects can be incorporated at appropriate spatial and temporal scales.

We have suggested a framework in which post-wildfire responses can be organized into post-wildfire response domains with three quantifiable metrics describing the range of characteristics for the fire, precipitation, and hydro-geomorphic regimes. By organizing the post-wildfire runoff and erosion responses into domains, this procedure will help to synthesize the data by identifying common patterns and generalities with the goal of understanding the reasons for different responses within and between domains.

Acknowledgments

This review is the product of numerous conversations with many interested specialists in science and land management. These conversations have spanned decades, and they span the continents. Some of the involved scientists are: Susana Bautista, Chris Chafer, Artemi Cerdà, Juan de la Fuente, Stefan Doerr, Brian Ebel, António Ferreira, Geoff Humphreys, Jason Kean, Patrick Lane, Isaac Larson, Lee MacDonald, Jorge Mataix-Solera, João Nunes, Petter Nyman, David Scott, Gary Sheridan, Hugh Smith, Dennis Staley, Cathelijne Stoof, Xavi Úbeda, Joe Wagenbrenner, Pete Wohlgemuth, and many others. Finally, we extend special thanks to two anonymous journal reviewers and Merche Bodí, Brian Ebel, Petter Nyman, Cathelijne Stoof, and Joe Wagenbrenner whose review comments encouraged us to 'dig' deeper and produce a more complete and focused review of the research issues facing the wildfire community.

References

- Agee, J.K., 1993. Fire Ecology of Pacific Northwest Forests. Island Press, Washington, D.C.(493 pp.).
- Anderson, H.W., Coleman, G.B., Zinke, P.J., 1959. Summer slides and winter scour; dry–wet erosion in southern California mountains. U.S. Department of Agriculture, Forest Service, Pacific Southwest Forest and Range Experiment Station, Research Paper, PSW 36. (12 pp.).
- Antoine, M., Javaux, M., Bielders, C.L., 2011. Integrating subgrid connectivity properties of the micro-topography in distributed runoff models, at the interrill scale. Journal of Hydrology 403, 213–223.
- Archibald, S., Bond, W.J., Stock, W.D., Fairbanks, D.H.K., 2005. Shaping the landscape: fire-grazer interactions in an African savanna. Ecological Applications 15, 96–109.
- Armanini, A., de Silvio, G., 1991. Fluvial hydraulics of mountain regions. Lecture Notes in Earth Sciences, vol. 37. Springer-Verlag, Heidelberg, Germany (468 pp.).

- ASRIS, 2012. Australian Soil Resource Information System. http://www.asris.csiro.au/ (accessed October 2012).
- Assouline, S., 2004. Rainfall-induced soil surface sealing: a critical review of observations, conceptual models, and solutions. Vadose Zone Journal 3, 570–591.
- Bachelet, D., Lenihan, J.M., Nielson, R.P., 2007. Wildfires and Global Climate Change. The Importance of Climate Change for Future Wildfire Scenarios in the Western United States, Excerpted From Ebi, Kristie, et al., Regional Impacts of Climate Change, Four Case Studies in the United States, Pew Center on Global Climate Change (20 pp. Available at http://www.pewclimate.org/docUploads/Regional-Impacts-West.pdf, last accessed 30 January 2012).
- Bachmann, J., Deurer, M., Arye, G., 2007. Modeling water movement in heterogeneous water-repellent soil: 1. Development of a contact angle dependent water-retention model. Vadose Zone Journal 6, 436–445.
- Baker, W.L., 2003. Fire and climate in forested landscapes of the U.S. Rocky Mountains. In: Veblen, T.T., Baker, W.L., Montenegro, G., Swetnam, T.W. (Eds.), Fire and Climatic Change in Temperate Ecosystems of the Western Americas. Springer, New York, pp. 120–157 (Chapter 5).
- Baker, W.L, Ehle, D., 2001. Uncertainty in surface-fire history: the case of ponderosa pine forest in the western United States. Canadian Journal of Forestry Research 31, 1205–1226.
- Banta, R.M., Schaaf, C.B., 1987. Thunderstorm genesis zones in the Colorado Rocky Mountains as determined by traceback of geosynchronous satellite images. Monthly Weather Review 115, 463–476.
- Barfield, B.J., Warner, R.C., Haan, C.T., 1981. Applied Hydrology and Sedimentology for Disturbed Areas. Oklahoma Technical Press, Stillwater, Oklahoma (145 pp.).
- Batchelor, G.K., 1982. The Theory of Homogeneous Turbulence. Cambridge University Press, New York (212 pp.).
- Beck, P.S.A., Goetz, S.J., Mack, M.C., Alexander, H.D., Jin, Y., Randerson, J.T., Loranty, M.M., 2011. The impacts and implications of an intensifying fire regime on Alaskan boreal forest composition and albedo. Global Change Biology 17. http://dx.doi.org/ 10.1111/j.1365-2486.2011.02412.x.
- Benavides-Solario, J., MacDonald, L.H., 2005. Measurement and prediction of post-fire erosion at the hillslope scale, Colorado Front Range. International Journal of Wildland Fire 14, 1–18.
- Benda, L., Dunne, T., 1997. Stochastic forcing of sediment supply to channel networks from landsliding and debris flow. Water Resources Research 33 (12), 2849–2863. Bento-Gonçalves, A., Vieira, A., Úbeda, X., Martin, D., 2012. Fire and soils: key concepts and
- recent advances. Geoderma. http://dx.doi.org/10.1016/j.geoderma.2012.01.004.
- Berg, N.H., Azuma, D.L., 2010. Bare soil and rill formation following wildfires, fuel reduction treatments, and pine plantations in the southern Sierra Nevada, California, USA. International Journal of Wildland Fire 19, 478–489.
- Berli, M., Chen, L., Young, M., 2008. Wildfire Effects on Watershed Hydrologic Processes: An Introduction for Hydraulic Engineers, Watershed Managers and Planners. Desert Research Institute (Publication No. 41243, 42 pp.).
- Bernardara, P., DeMichele, C., Rosso, R., Passoni, G., 2003. Space-Time Dynamics of Rainfall: Spectral Analysis, Proceedings of the 4th EGS Plinius Conference, Mallorca, Spain (4 pp.).
- Berner, L.T., Beck, P.A., Loranty, M.M., Alexander, H.D., Mack, M.C., Goetz, S.J., 2012. Cajander larch (*Larix cajanderi*) biomass distribution, fire regime and post-fire recovery in northeastern Siberia. Biogeosciences Discussions 9 (6), 7555–7600.
- Betson, R.P., 1964. What is watershed runoff? Journal of Geophysical Research 69, 1541-1552.
- Beven, K.J., 1996. Equifinality and uncertainty in geomorphological modeling. In: Rhoads, B.L., Thorn, C.E. (Eds.), The Scientific Nature of Geomorphology: Proceedings of the 27th Binghamton Symposium in Geomorphology, 27–29 September 1996. John Wiley & Sons Ltd., Chichester, England, pp. 289–313.
- Beven, K.J., 2000. Rainfall-Runoff Modeling-the Primer. John Wiley and Sons, Ltd., Chichester, England (353 pp.).
- Beven, K.J., Kirkby, M.J., 1979. A physically-based variable contributing area model of basin hydrology. Hydrological Sciences Bulletin 24, 43–69.
- Bigio, E., Swetnam, T.W., Baisan, C.H., 2010. A comparison and integration of tree-ring and alluvial records of fire history at the Missionary Ridge Fire, Durango, Colorado, USA. The Holocene 20 (7), 1047–1061.
- Blake, W.H., Wallbrink, P.J., Wilkinson, S.N., Humphreys, G.S., Doerr, S.H., Shakesby, R.A., Tomkins, K.M., 2009. Deriving hillslope sediment budgets in wildfire-affected forests using fallout radionuclide tracers. Geomorphology 104, 105–116.
- Bodi, M.B., Mataix-Solera, J., Doerr, S.H., Cerdà, A., 2011. The wettability of ash from burned vegetation and its relationship to Mediterranean plant species type, burn severity and total organic carbon content. Geoderma 160, 599–607.
- Bowman, D.S., Balch, J., Artaxo, P., Bond, W.J., Cochrane, M.A., D'Antonio, C.M., DeFries, R., Johnston, F.H., Keeley, J.E., Krawchuk, M.A., Kull, C.A., Mack, M., Moritz, M.A., Pyne, S., Roos, C.I., Scott, A.C., Sodhi, N.S., Swetnam, T.W., 2011. The human dimension of fire regimes on Earth. Journal of Biogeography 38 (12), 2223–2236.
- Brown, J.A.H., 1972. Hydrologic effects of a bushfire in a catchment in south-eastern New South Wales. Journal of Hydrology 15, 77–96.
- Brown, J.K., 2000. Introduction and fire regimes. In: Brown, J.K., Smith, J.K. (Eds.), Wildland fire in ecosystems—effects of fire on flora. General Technical Report RMRS-GTR-42-Volume 2. U.S. Department of Agriculture, Forest Service, Ogden, UT, pp. 1–8.
- Brown, J.R., Jakob, C., Haynes, J.M., 2010. An evaluation of rainfall frequency and intensity over the Australian Region in a global climate model. Journal of Climate 23, 6504–6525.
- Bryan, R.B., 2000. Soil erodibility and processes of water erosion on hillslope. Geomorphology 32, 385–415.
- Burns, J.L., 1953. Small-scale topographic effects on precipitation distribution in San Dimas Experimental Forest. Eos, Transactions American Geophysical Union 34 (1), 761–768.

Byram, G.M., Martin, R.E., 1970. The modeling of fire whirlwinds. Forest Science 16, 386–399.

- Cammeraat, E.L.H., 2004. Scale dependent thresholds in hydrological and erosion response of a semi-arid catchment in southeast Spain. Agriculture, Ecosystems and Environment 104, 317–332.
- Campbell, G.S., Jungbauer Jr., J.D., Bristow, K.L., Hungerford, R.D., 1995. Soil temperature and water content beneath a surface fire. Soil Science 159 (6), 363–374.
- Candela, A., Aronica, G., Santoro, M., 2005. Effects of forest fires on flood frequency curves in a Mediterranean catchment. Hydrological Sciences 50, 193–206.
- Cannon, S.H., Gartner, J.E., 2005. Wildfire-related debris flow from a hazards perspective. In: Hungr, O., Jakob, M. (Eds.), Debris-Flow Hazards and Related Phenomena. Springer-Praxis Books in Geophysical Sciences, Chichester, UK, pp. 321–344.
- Cannon, S.H., Bigio, E.R., Mine, E., 2001a. A process for fire-related debris flow initiation, Cerro Grande fire, New Mexico. Hydrological Processes 15, 3011–3023.
- Cannon, S.H., Kirkham, R.M., Parise, M., 2001b. Wildfire-related debris-flow initiation processes, Storm King Mountain, Colorado. Geomorphology 39, 171–188.
- Cannon, S.H., Gartner, J.E., Parrett, C., Parise, M., 2003. Wildfire-related debris flow generation through episodic progressive sediment bulking processes, western U.S.A. In: Rickenmann, D., Chen, C.L. (Eds.), Debris-Flow Hazards Mitigation–Mechanics, Prediction, and Assessment, Proceedings of the Third International Conference on Debris-Flow Hazards Mitigation, Davos, Switzerland, 10–12 September 2003. Balkema, Rotterdam, pp. 71–82.
 Cannon, S.H., Gartner, J.E., Wilson, R.C., Laber, J.L., 2008. Storm rainfall conditions for
- Cannon, S.H., Gartner, J.E., Wilson, R.C., Laber, J.L., 2008. Storm rainfall conditions for floods and debris flows from recently burned areas in southwestern Colorado and southern California. Geomorphology 96, 250–269.
- Cannon, S.H., Gartner, J.E., Rupert, M.G., Michael, J.A., Rea, A.H., Parrett, C., 2010. Predicting the probability and volume of post-wildfire debris flows in the intermountain west, USA. Geological Society of America Bulletin 122, 127–144.
- Cannon, S.H., Boldt, E.M., Kean, J.W., Laber, J., Staley, D.M., 2011. Rainfall intensity– duration thresholds for postfire debris-flow emergency-response planning. Natural Hazards 59, 209–236.
- Cawson, J., Sheridan, G., Lane, P., Smith, H., 2010. Characterising fire severity patches to understand how burn patchiness affects runoff connectivity and erosion. Geophysical Research Abstracts 12, EGU2010-3009.
- Cerdà, A., 1996. Seasonal variability of infiltration rates under contrasting slope conditions in southeast Spain. Geoderma 69, 217–232.
- Cerdà, A., 1998a. Changes in overland flow and infiltration after a rangeland fire in a Mediterranean scrubland. Hydrological Processes 12, 1031–1042.
- Cerdà, A., 1998b. Post-fire dynamics of erosional processes under Mediterranean climatic conditions. Zeitschrift fuer Geomorphologie Neue Folge 42, 373–398.
- Cerdà, A., Doerr, S.H., 2005. Influence of vegetation recovery on soil hydrology and erodibility following fire: an 11-year investigation. International Journal of Wildland Fire 14, 423–437.
- Cerdà, A., Lasanta, A., 2005. Long-term erosional responses after fire in the central Spanish Pyrenees: 1. Water and sediment yield. Catena 60, 59–80.
- Cerdà, A., Robichaud, P., 2009. Fire effects on soil infiltration. In: Cerdà, A., Robichaud, P. (Eds.), Fire Effects on Soils and Restoration Strategies. NH Science Publishers, Enfield, NH, pp. 81–104.
- Cerrelli, G.A., 2005. FIRE HYDRO, a simplified method for predicting peak discharges to assist in the design of flood protection measures for western wildfires. In: Moglen, G.E. (Ed.), Proceedings: 2005 Watershed Management Conference—Managing Watersheds for Human and Natural Impacts: Engineering, Ecological and Economic Challenges: 2005 July 19–22. American Society of Civil Engineers, Williamsburg, VA, pp. 935–941.
- Certini, G., 2005. Effects of fire on properties of forest soils: a review. Oecologia 143, 1-10.
- Chafer, C.J., 2008. A comparison of fire severity measures: an Australian example and implications for predicting major areas of soil erosion. Catena 74, 235–245.
- Chen, F., Warner, T.T., Manning, K., 2001. Sensitivity of orographic moist convection to landscape variability: a study of the Buffalo Creek, Colorado, flash flood case of 1996. Journal of the Atmospheric Sciences 58, 3204–3223.
- Coelho, C.O.A., Ferreira, A.J.D., Boulet, A.K., Keizer, J.J., 2004. Overland flow generation processes, erosion yields and nutrient loss under fires with different intensities lessons learned from analysis at different scales. Quarterly Journal of Engineering Geology & Hydrogeology 37, 233–240.
- Collins, L.M., Ketcham, B., 2005. Fluvial geomorphic response of a Northern California coastal stream to wildfire. U.S. Department of the Interior, National Park Service Vision Fire Lessons Learned from the October 1995 Fire. Point Reyes National Seashore, pp. 59–79.
- Conedera, M., Peter, L., Marxer, P., Forster, F., Rickenmann, D., Re, L., 2003. Consequences of forest fires on the hydrogeological response of mountain catchments: a case study of the Riale Buffaga, Ticino, Switzerland. Earth Surface Processes and Landforms 28, 117–129.
- Costa, J.E., 1987. A comparison of the largest rainfall-runoff floods in the United States with those of the People's Republic of China and the world. Journal of Hydrology 96, 101–115.
- Costa, J.E., 1988. Rheologic, geomorphic, and sedimentologic differentiation of water floods, hyperconcentrated flows, and debris flow. In: Baker, V.R., Kochel, R.C., Patten, P.C. (Eds.). Flood Geomorphology. Wiley-Interscience. New York. pp. 113–122.
- Cowpertwait, P.S.P., Lockie, T., Davis, M.D., 2004. A stochastic spatial-temporal disaggregation model for rainfall. Research Letters in the Information and Mathematical Sciences 6, 109–122.
- Daly, C., Neilson, R.P., Phillips, D.L., 1994. A statistical-topographic model for mapping climatological precipitation over mountainous terrain. Journal of Applied Meteorology 33, 140–158.

- Davenport, D.W., Breshears, D.D., Wilcox, B.P., Allen, C.D., 1998. Viewpoint: sustainability of piñon-juniper ecosystems-a unifying perspective of soil erosion thresholds. Journal of Range Management 51 (2), 231–240.
- DeBano, L.F., 1981. Water repellent soils: a state-of-the-art. U.S. Department of Agriculture, Forest Service General Technical Report PSW-46. (21 pp.).
- DeBano, L.F., 2000. The role of fire and soil heating on water repellency in wildland environments: a review. Journal of Hydrology 231–232, 195–206.
- DeBano, L.F., Savage, S.M., Hamilton, D.A., 1976. The transfer of heat and hydrophobic substances during burning. Soil Science Society of America Journal 40 (5), 779–782. DeBano, L.F., Neary, D.G., Ffolliott, P.F., 1998. Fire's Effects on Ecosystems. John Wiley &
- Sons, New York 159–196. Decagon, 2006. Mini Disk Infiltrometers User's Manual. Decagon Devices, Pullman, WA
- (18 pp.).
- Dekker, L.W., Ritsema, C.J., 2000. Wetting patterns and moisture variability in water repellent Dutch soils. Journal of Hydrology 231–232, 148–164. Desilets, S.L.E., Nigssen, B., Ekwurzel, B., Ferré, T.P.A., 2007. Post-wildfire changes in
- Desilets, S.L.E., Nigssen, B., Ekwurzel, B., Ferré, T.P.A., 2007. Post-wildfire changes in suspended sediment rating curves: Sabino Canyon, Arizona. Hydrological Processes 21, 1413–1423.
- Díaz-Delgado, R., Lloret, F., Pons, X., 2004. Statistical analysis of fire frequency models for Catalonia (NE Spain), 1975–1998 based on fire scar maps from Landsat MSS data. International Journal of Wildland Fire 13, 89–99.
- Dick, G.S., Anderson, R.S., Sampson, D.E., 1997. Controls on flash flood magnitude and hydrograph shape, Upper Blue Hills badlands, Utah. Geology 25 (1), 45–48.
- Dietrich, W.E., Wilson, C.J., Montgomery, D.R., McKean, J., Bauer, R., 1992. Erosion thresholds and land surface morphology. Geology 20, 675–679.
- Doehring, D.O., 1968. The effect of fire on geomorphic processes in the San Gabriel Mountains, California. Contributions to Geology 7, 43–65.
- Doerr, S.H., Moody, J.A., 2004. Hydrological effects of soil water repellency: on spatial and temporal uncertainties. Hydrological Processes 18, 829–832.
- Doerr, S.H., Shakesby, R.A., Walsh, R.P.D., 2000. Soil water repellency: its causes, characteristics and hydro-geomorphological significance. Earth-Science Reviews 51, 33–65.
- Doerr, S.H., Shakesby, R.A., MacDonald, L.H., 2009. Soil water repellency: a key factor in post-fire erosion. In: Cerdà, A., Robichaud, P.R. (Eds.), Fire Effects on Soils and Restoration Strategies. Science Publishers, New Hampshire, Enfield, NH, pp. 197–223.
- D'Odorico, P.R., Rigon, R., 2003. Hillslope and channel contributions to the hydrologic response. Water Resources Research 39 (5), 1–9.
- Dunkerley, D.L., 2010. How do the rain rates of sub-event intervals such as the maximum 5 and 15-min rates (I₅ or I₃₀) relate to the properties of the enclosing rainfall event. Hydrological Processes 24, 2425–2439.
- Dunkerley, D.L., 2011. Effects of rainfall intensity fluctuations on infiltration and runoff: rainfall simulation on dryland soils, Fowlers Gap, Australia. Hydrological Processes 26 (15), 2211–2224.
- Dunkerley, D., Martin, N., Berg, S., Ferguson, R., 2009. Fire, catchment runoff and erosion processes and post-fire rehabilitation programs: recent Australian experience. In: Cerdà, A., Robichaud, P.R. (Eds.), Fire Effects on Soils and Restoration Strategies. Science Publishers, Enfield, NH, pp. 467–509.
- Dunne, T., 1978. Field studies of hillslope flow processes. In: Kirkby, M.J. (Ed.), Hillslope Hydrology. John Wiley & Sons, New York, pp. 227–293.
- Ebel, B.A., Moody, J.A., 2012. Rethinking infiltration in wildfire-affected soils. Hydrological Processes. http://dx.doi.org/10.1012/hyp.9696.
- Ebel, B.A., Moody, J.A., Martin, D.A., 2012. Hydrologic conditions controlling runoff generation immediately after wildfire. Water Resources Research 48. http://dx.doi.org/ 10.1029/2011WR011470 W03529.
- Eagleson, F.S., Shack, W.J., 1966. Some criteria for the measurement of rainfall and runoff. Water Resource Research 2 (3), 427–436.
- Elliot, W.J., Hall, D.E., 2010. Disturbed WEPP Model 2.0. Ver. 2011.11.22. U.S. Department of Agriculture, Forest Service, Rocky Mountain Research Station, Moscow, Idaho (Online at http://forest.moscowfsl.wsu.edu/fswepp).
- Elliot, W.J., Liebenow, A.M., Laflen, J.M., Kohl, K.D., 1989. A Compendium of Soil Erodibility Data Form WEPP Cropland Soil Field Erodibility Experiments 1987 & 88. NSERL Report No. 3.Ohio State University & USDA Agricultural Research Service (Part A, 29 pp.; Part B, 291 pp.).
- Elliot, W.J., Scheele, D.L., Hall, D.E., 1999. Rock: Clime-Rocky Mountain Research Station Climate Generator. Moscow, Idaho. U.S. Department of Agriculture, Forest Service, Rocky Mountain Research Station, Moscow Forestry Science Laboratory (Online at <<u>http://forest.moscowfsl.wsu.edu/fswepp></u>).
- Elliot, W.J., Hall, D.E., Robichaud, P.R., 2010. FS Peak Flow Calculator. Version 2010.10.28. U.S. Department of Agriculture, Forest Service, Rocky Mountain Research Station, Moscow, ID (online at http://forest.moscowfsl.wsu.edu/fswepp/ermit/peakflow, last accessed 30 January 2012).
- Etheredge, D., Gutzler, D.S., Pazzaglia, F.J., 2004. Geomorphic response to seasonal variations in rainfall in the southwest United States. Geological Society of America Bulletin 116, 606–618.
- European Commission, 2012. Forest Fires in Europe, Middle East and North Africa. EUR 25483 EN. Publications Office of the European Union, Luxembourg (109 pp.).
- Feldman, A.D., 2000. HEC-1 Flood Hydrograph Package. In: Singh, V.P. (Ed.), Computer Models of Watershed Hydrology. Water Resources Publications, Highlands Ranch, CO, pp. 119–150.
- Ferreira, A.J.D., Coelho, C.O.A., Bolet, A.K., Leighton-Boyce, G., Keizer, J.J., Ritsema, C.J., 2005. Influence of burning intensity on water repellency and hydrological processes at forest and shrub sites in Portugal. Australian Journal of Soil Research 43, 327–336.
- Fiedler, F.R., Ramirez, J.A., 2000. A numerical method for simulating discontinuous shallow flow over an infiltrating surface. International Journal for Numerical Methods in Fluids 32, 219–240.

- Finley, C.D., Glenn, N.F., 2010. Fire and vegetation type effects on soil hydrophobicity and infiltration in the sagebrush-steppe: II. Hyperspectral analysis. Journal of Arid Environments 74 (6), 660–666.
- Finnerty, B.D., Smith, M.B., Koren, V., Seo, D.J., Moglen, G., 1997. Space-time scale sensitivity of the Sacramento Model to radar-gage precipitation inputs. Journal of Hydrology 203, 21–38.
- Flanagan, D.C., Nearing, M.A., 1995. USDA-Water Erosion Prediction Project, hillslope profile and watershed model documentation. NSERL Report No. 10.USDA-ARS National Soil Erosion Research Laboratory. West Lafavette. Indiana. USA (298 pp.).
- National Soil Erosion Research Laboratory, West Lafayette, Indiana, USA (298 pp.). Flannigan, M.D., Stocks, B.J., Wotton, B.M., 2000. Climate change and forest fires. Science of the Total Environment 262, 221–229.
- Florsheim, J.L., Keller, E.A., Best, D.W., 1991. Fluvial sediment transport in response to moderate storm flows following chaparral wildfire, Ventura County, southern California. Geological Society of America Bulletin 103, 504–511.
- Foltz, R.B., Robichaud, P.R., Rhee, H., 2009. A synthesis of post-fire road treatments for BAER teams: methods, treatment effectiveness, and decision making tools for rehabilitation. General Technical Report, RMRS-GTR-228.U.S. Department of Agriculture, Forest Service, Rocky Mountain Research Station, Fort Collins, CO (152 pp.).
- Foster, G.R., Meyer, L.D., 1972. A closed-form soil erosion equation for upland areas. In: Shen, H.W. (Ed.), Sedimentation: Symposium to Honor Professor H.A. Einstein. Colorado State University, Fort Collins, CO, pp. 12.1–12.19.
- Foster, G.R., Meyer, L.D., Onstad, C.A., 1977. An erosion equation derived from basic erosion principles. Transactions of ASAE 20 (4), 678–682.
- Foster, G.R., McCool, D.K., Renard, K.G., Moldenhauer, W.C., 1981. Conversion of the universal soil loss equation to SI metric units. Journal of Soil and Water Conservation 36 (6), 355–359.
- Foster, G.R., Flanagan, D.C., Nearing, M.A., Lane, L.J., Risse, L.M., Finkner, S.C., 1995. Hillslope erosion component, chap. 11. In: Flanagan, D.C., Nearing, M.A. (Eds.), USDA-Water Erosion Prediction Project, Hillslope profile and watershed Model Documentation, NSERL Report no. 10, West Layfayette, Indiana, pp. 11-1-11-2.
- Foufoula-Georgiou, E., Krajewski, W., 1995. Recent advances in rainfall modeling, estimation and forecasting. U.S. National Report to International Union Geodesy and Geophysics 1991–1994: Reviews of Geophysics, Supplement, pp. 1125–1137.
- Fox, D.M., Darboux, F., Carrega, P., 2007. Effects of fire-induced water repellency on soil aggregate stability, splash erosion, and saturated hydraulic conductivity for different size fractions. Hydrological Processes 21, 2377–2384.
- Gabet, E.J., 2003a. Post-fire thin debris flows: sediment transport and numerical modelling. Earth Surface Processes and Landforms 28, 1341–1348.
- Gabet, E.J., 2003b. Sediment transport by dry ravel. Journal of Geophysical Research 108, 2049. http://dx.doi.org/10.1029/2001JB001686 (8 pp.).
- Gabet, E.J., Dunne, T., 2003. Sediment detachment by rain power. Water Resources Research 39 (1), 1-1–1-12. http://dx.doi.org/10.1029/2001WR000656.
- Gabet, E.J., Mudd, S.M., 2006. The mobilization of debris flows from shallow landslides. Geomorphology 74, 207–218.
- Gabet, E.J., Sternberg, P., 2008. The effects of vegetative ash on infiltration capacity, sediment transport, and the generation of progressively bulked flows. Geomorphology 101, 666–673.
- Gallaway, J.M., Martin, Y.E., Johnson, E.A., 2009. Sediment transport due to tree root throw: integrating tree population dynamics, wildfire and geomorphic response. Earth Surface Processes and Landforms 34 (9), 1255–1269.
- García-Ruiz, J.M., Arnáez, J., Gómez-Villar, A., Ortígosa, L., Lana-Renault, N., 2012. Firerelated debris flows in the Iberian Range, Spain. Geomorphology. http://dx.doi.org/ 10.1016/j.geomorph.2012.03.032.
- Gartner, J.E., Bigio, E.R., Cannon, S.H., 2004. Compilation of post wildfire runoff-event data from the western United States. (online USGS Open-File Report 2004–1085, http://pubs.usgs.gov/of/2004/1085 [accessed: 14 Jan 2004]).
- Gartner, J.E., Cannon, S.H., Santi, P.M., deWolf, V.G., 2008. Empirical models to predict debris flow volumes generated from recently burned basins in the western U.S Geomorphology 96, 339–354.
- Germanoski, D., Miller, J.R., Latham, D., 2002. The importance of event sequencing on the geomorphic impact of wildfire in the central Great Basin. Geological Society of America, Abstracts With Programs 34, 319.
- Germer, K., Braun, J., 2011. Effects of saturation on slope stability: laboratory experiments utilizing external load. Vadose Zone Journal 10, 447–486.
- Giovannini, G., Lucchesi, S., 1983. Effect of fire on hydrophobic and cementing substances of soil aggregates. Soil Science 136, 231–236.
- Giovannini, G., Lucchesi, S., Giachetti, M., 1988. Effects of heating on some physical and chemical parameters related to soil aggregation and erodibility. Soil Science 146, 255–261.
- Goforth, B.R., Graham, R.C., Hubbert, K.R., Zanner, C.W., Minnich, R.A., 2005. Spatial distribution and properties of ash and thermally altered soils after high-severity forest fire southern California. International Journal of Wildland Fire 14, 343–354. Gomez, B., 1991. Bedload transport. Earth-Science Reviews 31, 89–132.
- González-Bororino, G., Osterkamp, W.R., 2004. Applying RUSLE 2.0 on burned-forest lands: an appraisal. Journal of Soil and Water Conservation 59 (1), 36–42.
- Goode, J.R., Luce, C.H., Buffington, J.M., 2012. Enhanced sediment delivery in a changing climate in semi-arid mountain basins: implications for water resource management and aquatic habitat in the northern Rocky Mountains. Geomorphology 139–140, 1–15.
- Green, W.A., Ampt, G.A., 1911. Studies on soil physics: 1. The flow of air and water through soils. Journal of Agriculture Science 4, 1–24.
- Grell, G.A., Dudhia, J., Stauffer, D.R., 1994. A description of the fifth-generation Penn State/NCAR Mesoscale Model (MM5). NCAR Technical Note, NCAR/TN-398 + STR.Mesoscale and Microscale Meteorology Division, National Center for Atmospheric Research, Boulder, CO.

- Gyssel, G., Poesen, J., Bochet, E., Li, Y., 2005. Impact of plant roots on the resistance of soils to erosion by water: a review. Progress in Physical Geography 29, 189–217.
- Hairsine, P.B., Rose, C.W., 1992. Modeling water erosion due to over land flow using physical principles: 2. Rill flow. Water Resources Research 28 (1). http://dx.doi.org/ 10.1029/91WR02381.
- Hanshaw, M.N., Schmidt, K.M., Jorgensen, D.P., Stock, J.D., 2008. By air and land: estimating post-fire debris-flow susceptibility through high-resolution radar reflectivity and tipping-bucket gage rainfall. EOS Transactions AGU, 2008 Fall Meeting Supplement, Abstract H51D-0850.
- Hardy, C.C., Menakis, J.P., Long, D.G., Brown, J.K., 1998. Mapping historic fire regimes for the Western United States: integrating remote sensing and biophysical data. In: Greer, J.D. (Ed.), Proceedings of the 7th Forest Service Remote Sensing Applications Conference; 1998 April 6–10, Nassau Bay, TX. Bethesda, MD, American Society of Photogrammetry and Remote Sensing, pp. 288–300.
- Harris, D., Menabde, M., Seed, A., Austin, G., 1998. Breakdown coefficients and scaling properties of rain fields. Nonlinear Processes in Geophysics 5, 93–104.
- Hassanizadeh, S.M., Celia, M.A., Dahle, H.K., 2002. Dynamic effect in the capillary pressure-saturation relationship and its impacts on unsaturated flow. Vadose Zone Journal 1, 38–57.
- Hawkins, R.H., 1973. Improved prediction of storm runoff in mountain watersheds. Journal of Irrigation and Drainage Division, Proceedings of the American Society of Civil Engineers 99, 519–523.
- Hawkins, R.H., 1982. Interpretations of source area variability in rainfall-runoff relations. International Symposium of Rainfall Runoff Modeling. Water Resources Publication, Littleton, CO, pp. 303–324.
- Hawkins, R.H., 1993. Asymptotic determination of runoff curve numbers from data. Journal of Irrigation and Drainage Engineering 119 (2), 334–345.
- Hawkins, R.H., Greenberg, R.J., 1990. WILDCAT4 Flow Model. School of Renewable Natural Resources. University of Arizona, Tucson, AZ.
- HEC-RAS, 2012. User manuals. available at http://www.hec.usace.army.mil/.
- Heede, B.H., Harvey, M.D., Laird, J.R., 1988. Sediment delivery linkages in a chaparral watershed following wildfire. Environmental Management 12 (3), 349–358.
- Heinselman, M.L., 1981. Fire intensity and frequency as factors in the distribution and structure of northern ecosystems. Fire Regimes and Ecosystem Properties: Proceedings of the Conference; 1978 December 11–15; Honolulu. General Technical Report WO-26. U.S. Department of Agriculture, Forest Service, Washington, D.C., pp. 7–57.
- Helvey, J.D., 1980. Effects of a north central Washington wildfire on runoff and sediment production. Water Resources Bulletin 16 (4), 627–634.
- Henz, J.F., 1973. Characteristics of severe convective storms on Colorado's high plains. Preprint for Eighth Conference on Severe Local Storms, American Meteorological Society, October 15–17 1973, Denver, CO, pp. 96–103.
- Hershfield, D.M., 1961. Rainfall frequency atlas of the United States for duration from 30 minutes to 24 hours and return periods from 1 to 100 years. U.S. Department of Commerce, Technical Paper No. 40. (107 pp.).
- Heyerdahl, E.K., Alvarado, E., 2003. Influence of climate and land use on historical surface fires in pine-oak forests, Sierra Madre Occidental, Mexico. In: Veblen, T.T., Baker, W.L., Montenegro, G., Swetnam, T.W. (Eds.), Fire and Climatic Change in Temperate Ecosystems of the Western Americas. Springer, New York, pp. 196–217.
- Hillel, D., 1998. Environmental Soil Physics. Academic Press, New York 385–426.
- Horton, R.E., 1939. Analysis of runoff-plot experiments with varying infiltration capacity. Eos, Transactions American Geophysical Union 20, 693–711 (Part IV).
- Huff, F.A., 1967. Time distribution of rainfall in heavy storms. Water Resources Research 3 (4), 1007–1019.
- Huffman, E.L., MacDonald, L.H., Stednick, J.D., 2001. Strength and persistence of fire induced soil hydrophobicity under ponderosa and lodgepole pine, Colorado Front Range. Hydrological Processes 15, 2877–2892.
- Humphreys, G.S., Craig, F.G., 1981. Effects of fire on soil chemical, structural and hydrological properties. In: Gill, A.M. (Ed.), Fire and the Australian Biota. Australian Academy of Science, Canberra, Australia, pp. 177–200.
- Hungerford, R.D., Frandsen, W.H., Ryan, K.C., 1996. Heat transfer into the duff and organic soil. U.S. Department of Agriculture, Forest Service, Intermountain Research Station, Missoula, Montana, Final Project Report, FWS Agreement No. 14-48-0009-92-962.
- Imeson, A.C., Verstraten, J.M., van Mullegen, E.J., Sevink, J., 1992. The effects of fire and water repellency on infiltration and runoff under Mediterranean type forest. Catena 19, 345–361.
- Inbar, M., Tamir, M., Wittenberg, L., 1998. Runoff and erosion processes after a forest fire in Mount Carmel, a Mediterranean area. Geomorphology 24, 17–33.
- Institution of Engineers, Australia, 1998. Australian Rainfall and Runoff: a Guide to Flood Estimation. In: Pilgrim, D.H. (Ed.), (Reprinted edition, Barton, ACT).
- Interagency Advisory Committee on Water Data, 1981. Guideline for determining flood flow frequency. U.S. Geological Survey Bulletin, 17B (28 pp.).
- Iverson, R.M., 1997. The physics of debris flows. Reviews of Geophysics 35, 245–296.
- Jackson, M., Roering, J.J., 2009. Post-fire geomorphic response in steep, forested landscapes: Oregon Coast Range, USA. Quaternary Science Reviews 28, 1131–1146.
- Jarrett, R.D., England, J.F. Jr, 2002. Reliability of paleostage indicators for paleoflood studies. Ancient Floods, Modern Hazards: Principles and Applications of Paleoflood Hydrology, Water Science and Application. American Geophysical Union, 5 91–109.
- Jarrett, A.R., Fritton, D.D., 1978. Effects of entrapped soil air on infiltration. Transactions of ASAE 21, 901–906.
- Jiang, Y., Zhuang, Q., Flannigan, M.D., Little, J.M., 2009. Characterization of wildfire regimes in Canadian boreal terrestrial ecosystems. International Journal of Wildland Fire 18, 992–1002.
- Johansen, M.P., Hakonson, T.E., Breshears, D.D., 2001. Post-fire runoff and erosion from rainfall simulation: contrasting forests with shrublands and grasslands. Hydrological Processes 15, 2953–2965.

- Jorgensen, D.P., Hanshaw, M.N., Schmidt, K.M., Laber, J.L., Staley, D.M., Kean, J.W., 2011. Value of a dual-polarized gap-filling radar in support of southern California post-fire debris-flow warnings. Journal of Hydrometeorology. http://dx.doi.org/10.1175/JHM-D-11-05-1.
- Karunarathna, A.K., Moldrup, P., Kawamoto, K., de Jonge, L.W., Komatsu, T., 2010. Tworegion model for soil water repellency as a function of matric potential and water content. Vadose Zone Journal 9, 719–730.
- Kean, J.W., Staley, D.M., Cannon, S.H., 2011. In situ measurements of post-fire debris flows in southern California: Comparisons of the timing and magnitude of 24 debris-flow events with rainfall and soil moisture conditions. Journal of Geophysical Research 116. http://dx.doi.org/10.1029/2011 [F002005 (F04019).
- Keeley, J.E., 2009. Fire intensity, fire severity and burn severity: a brief review and suggested usage. International Journal of Wildland Fire 18, 116–126.
- Keeley, J.E., Brennan, T., Pfaff, A.H., 2008. Fire severity and ecosystem responses following crown fires in California shrublands. Ecological Applications 18, 1530–1546.
- Keizer, J.J., Coelho, C.O.A., Shakesby, R.A., Domingues, C.S.P., Malvar, M.C., Perez, I.M.P., Matias, M.J.S., Ferreira, A.J.D., 2005. The role of soil water repellency in overland flow generation in pine and eucalypt forest stands in coastal Portugal. Australian Journal of Soil Research 43, 337–349.
- Key, C.H., Benson, N.C., 2004. Landscape assessment: remote sensing of severity, the normalized burn ratio; and ground measure of severity, the composite burn index. In: Lutes, D.C., Keane, R.E., Caratti, J.F., Key, C.H., Benson, N.C., Gangi, L.J. (Eds.), FIREMON: Fire Effects Monitoring and Inventory System, Ogden, UT, US Department of Agriculture, Forest Service, Rocky Mountain Research Station General Technical Report RMRS-GTR-164-CD; LA-1-55.
- Kharuk, V.I., Dvinskaya, M.L., Ranson, K.J., 2012. Fire return intervals within the northern boundary of the larch forest in Central Siberia. International Journal of Wildland Fire 22 (2), 207–211.
- King, P.M., 1981. Comparison of methods for measuring severity of water repellence of sandy soils and assessment of some factors that affect its measurement. Australian Journal of Soil Research 19, 275–285.
- Kinner, D.A., Moody, J.A., 2008. Infiltration and runoff measurements on steep burned hillslopes using a rainfall simulator with variable rain intensities. U.S. Geological Survey Scientific Investigations Report 2007–5211. (64 pp.).
- Kinner, D.A., Moody, J.A., 2010. Spatial variability of steady-state infiltration into a twolayer soil system on burned hillslopes. Journal of Hydrology 381, 322–332.
- Kirkby, M.J., 1976. Test of the random network model and its application to basin hydrology. Earth Surface Processes 1, 197–212.
- Kirkby, M.J., 2011. A personal synthesis and commentary, hydro-geomorphology, erosion and sedimentation. In: Kirkby, M.J. (Ed.), Benchmark Papers in Hydrology, 6. International Association of Hydrological Science, IAHS, pp. 1–36.
- Kitzberger, T., Veblen, T.T., 2003. Influences of climate on fire in northern Patagonia. In: Veblen, T.T., Baker, W.L., Montenegro, G., Swetnam, T.W. (Eds.), Fire and Climatic Change in Temperate Ecosystems of the Western Americas. Springer, New York, pp. 296–321.
- Kokaly, R.F., Rockwell, B.W., Haire, S.L., King, T.V.V., 2007. Characterization of post-fire surface cover, soils, and burn severity at the Cerro Grande Fire, New Mexico, using hyperspectral and multispectral remote sensing. Remote Sensing of Environment 106, 305–325.
- Krammes, J.S., 1965. Seasonal debris movement from steep mountainside slopes in southern California. Federal Inter-Agency Sedimentation Conference, Miscellaneous Publication U.S. Department of Agriculture, 970, pp. 85–88.
- Krammes, J.S., DeBano, L.F., 1965. Soil wettability: a neglected factor in watershed management. Water Resources Research 1 (2), 283–286.
- Krammes, J.S., Osborn, J., 1969. Water-repellent soils and wetting agents as factors influencing erosion. In: Debano, L.F., Letey, J. (Eds.), Water-repellent Soils, Proceedings of the Symposium on Water-Repellent Soils. University of California, Riverside, CA, pp. 177–187.
- Krawchuk, M.A., Moritz, M.A., 2009. Fire regimes of China: inference from statistical comparison with the United States. Global Ecology and Biogeography 18, 626–639.
- Krebs, P., Pezzatti, G.B., Massoleni, S., Talbot, L.M., Conedera, M., 2010. Fire regimes: history and definition of a key concept in disturbance ecology. Theory in Biosciences 129, 53–69.
- Kunze, M.D., Stednick, J.D., 2006. Streamflow and suspended sediment yield following the 2000 Bobcat Fire, Colorado. Hydrological Processes 20, 1661–1681.
- Kutílek, M., 1980. Constant-rainfall infiltration. Journal of Hydrology 45, 289-303.
- Lamb, M.P., Dietrich, W.E., Venditti, J.G., 2008. Is the critical Shields stress for incipient sediment motion dependent of channel-bed slope? Journal of Geophysical Research 113. http://dx.doi.org/10.1029/2007/JF000831.
- Lamb, M.P., Scheingross, J.S., Amidon, W.H., Swanson, E., Limaye, A., 2011. A model for fire-induced sediment yield by dry ravel in steep landscapes. Journal of Geophysical Research 116 (F03006).
- Lane, P.N.J., Sheridan, G.J., Noske, P.J., 2006. Changes in sediment loads and discharge from small mountain catchments following wildfire in south eastern Australia. Journal of Hydrology 331, 495–510.
- Lane, P.N.J., Sheridan, G.J., Noske, P.J., Sherwin, C.B., Costenaro, J.L., Nyman, P., Smith, H.G., 2012. Fire effects on forest hydrology: lessons from a multi-scale catchment experiment in SE Australia, IAHS Publ. 353. Revisiting Experimental Catchment Studies in Forest Hydrology: Proceeding of a Workshop Held During the XXV IUGG General Assembly in Melbourne, June–July 2011, pp. 137–143. Lanini, J.S., Clark, E.A., Lettenmaier, D.P., 2009. Effects of precipitation-fire timing
- Lanini, J.S., Clark, E.A., Lettenmaier, D.P., 2009. Effects of precipitation-fire timing and regime of post-fire sediment delivery in Pacific Northwest forest. Geophysical Research Letters 36, L01402. http://dx.doi.org/10.1029/2008GL034588.
- Larsen, I.J., 2003. From the rim to the river—the geomorphology of debris flows in the Green River Canyons of Dinosaur National Monument, Colorado and Utah. M.S. Thesis, Utah State University, Logan UT, 196 pp.

- Larsen, I.J., MacDonald, L.H., Brown, E., Rough, D., Welsh, M.J., Pietraszek, J.H., Libohova, Z., Benavides-Solorio, J., Schaffrath, K., 2009. Causes of post-fire runoff and erosion: roles of soil water repellency, surface cover, and soil sealing? Soil Science Society of America Journal 73, 1393–1407.
- Lawrence, D.S.L., 1997. Macroscale surface roughness and frictional resistance in overland flow. Earth Surface Processes and Landforms 22, 365–382.
- Leavesley, G.H., Hay, L., 1998. The use of coupled atmospheric and hydrological models for water-resources management in headwater basins. In: Kovar, K., Tappeiner, U., Peters, N.E., Craig, R.G. (Eds.), Hydrology, Water Resources and Ecology in Headwaters: IAHS Publication No. 248, pp. 259–265.
- Lehmann, P., Hinz, C., McGrath, G., Tromp-van Meerveld, H.J., McDonnell, J.J., 2007. Rainfall threshold for hillslope outflow: an emergent property of flow pathway connectivity. Hydrology and Earth System Sciences 11, 1047–1063.
- Letey, J., Carrillo, M.L.K., Pang, X.P., 2000. Approaches to characterize the degree of water repellency. Journal of Hydrology 231–232, 61–65.
- Lewis, S.A., Robichaud, P.R., Frazier, B.E., Wu, J.Q., Laes, D.Y.M., 2008. Using hyperspectral imagery to predict post-wildfire soil water repellency. Geomorphology 95, 192–205. Linsley, R.K., 1958. Correlation of rainfall intensity and topography in northern California.
- Eos, Transactions American Geophysical Union 39 (1), 15–18. Littell, J.S., McKenzie, D., Peterson, D.L., Westerling, A.L., 2009. Climate and wildfire
- area burned in western U.S. ecoprovinces, 1916–2003. Ecological Applications 19, 1003–1021.
- Loomis, J., Wohlgemuth, P., González-Cabán, A., 2003. Economic benefits of reducing fire-related sediment in southwestern fire-prone ecosystems. Water Resources Research 39 (8). http://dx.doi.org/10.1029/2003WR002176.
- Lu, H., Gallant, J., Prosser, I.P., Moran, C., Priestley, G., 2001. Prediction of sheet and rill erosion over the Australian continent, incorporating monthly soil loss distribution. Technical Report 13/01.CSIRO Land and Water, Canberra, Australia Available at: http://www.clw.csiro.au/publications/technical2001/tr13-01.pdf.
- Malamud, B.D., Millington, J.D.A., Perry, G.L.W., 2005. Characterizing wildfire regimes in the United States. Proceedings of the National Academy of Sciences of the United States of America 102 (13), 4694–4699.
- Mallik, A.U., Gimingham, C.H., Rahman, A.A., 1984. Ecological effects of Heather burning, I. Water infiltration, moisture retention and porosity of surface soil. Journal of Ecology 72, 767–776.
- Martin, D.A., Moody, J.A., 2001. Comparison of soil infiltration rates in burned and unburned mountainous watersheds. Hydrological Processes 15, 2893–2903.
- Martin, Y., Valeo, C., Tait, M., 2008. Centimetre-scale digital representations of terrain and impacts on depression storage and runoff. Catena 75 (2), 223–233.
- Mataix-Solera, J., Cerdà, A., Arcenegui, V., Jordán, A., Zavala, L.M., 2011. Fire effects on soil aggregation: a review. Earth-Science Reviews 109, 44–60.
- Matrosov, S.Y., Cifelli, R., Gochis, D., 2013. Measurements of heavy convective rainfall in the presence of hail in flood-prone areas using an X-ban polarimetric radar. Journal of Applied Meteorology and Climatology 52, 395–407.
- Mayor, A.G., Bautista, S., Llovet, J., Bellot, J., 2007. Post-fire hydrological and erosional responses of a Mediterranean landscape: seven years of catchment-scale dynamics. Catena 71, 68–75.
- Mayor, A.G., Bautista, S., Small, E.E., Dixon, M., Bellot, J., 2008. Measurement of the connectivity of runoff source areas as determined by vegetation pattern and topography: a tool for assessing potential water and soil losses in drylands. Water Resources Research 44, W10423. http://dx.doi.org/10.1029/2007WR006367 (13 pp.).
- McCoy, S.W., Kean, J.W., Coe, J.A., Tucker, G.E., Staley, D.M., Wasklewicz, T.A., 2012. Sediment entrainment by debris flows: in situ measurements from the headwater of a steep catchment. Journal of Geophysical Research 117. http://dx.doi.org/ 10.1029/2011JF002278 (25 pp.).
- Mein, R.G., Larson, C.L., 1973. Modeling infiltration during a steady rain. Water Resources Research 9 (2), 384–394.
- Menabde, M., Harris, D., Seed, A.W., Austin, G.L., Stow, C.D., 1997. Multiscaling properties of rainfall and bounded random cascades. Water Resources Research 33 (12), 2823–2830.
- Menéndez-Duarte, R., Fernández, S., Soto, J., 2009. The application of ¹³⁷Cs to post-fire erosion in north-west Spain. Geoderma 150, 54–63.
- Meyer, G.A., Wells, S.G., 1997. Fire-related sedimentation events on alluvial fans, Yellowstone National Park, U.S.A Journal of Sedimentary Research 67, 776–791.
- Meyer, G.A., Wells, S.G., Jull, A.J.T., 1995. Fire and alluvial chronology in Yellowstone National Park: climatic and intrinsic controls on Holocene geomorphic processes. Geological Society of America Bulletin 107 (10), 1211–1230.
- Miller, J.F., Frederick, R.H., Tracey, R.J., 1973. Precipitation-Frequency Atlas of the Western United States, Colorado, NOAA Atlas 2, vol. III National Weather Service (67 pp.).
- Milly, P.C.D., Wetherald, R.T., 2002. Macroscale water fluxes 3. Effects of land processes on the variability of monthly river discharge. Water Resources Research 38 (11), 17-1–17-12. http://dx.doi.org/10.1029/2001WR000761.
- Milzow, C., Kinzelbach, W.K., 2010. Accounting for subgrid scale topographic variations in flood propagation modeling using MODFLOW. Water Resources Research 46. http://dx.doi.org/10.1029/2009WR008088 W10521.
- Mockus, V., 1972. Estimation of Direct Runoff From Storm Rainfall. NRCS National Engineering Handbook, Section 4, Chapter 10. U.S. Department of Agriculture, Washington, D.C. 10-1–10-24.
- Montes-Helu, M.C., Kolb, T., Dore, S., Sullivan, B., Hart, S.C., Koch, G., Hungate, B.A., 2009. Persistent effects of fire-induced vegetation change on energy partitioning and evapotranspiration in ponderosa pine forests. Agricultural and Forest Meteorology 149 (3–4), 491–500.

Montgomery, D.R., Dietrich, W.E., 1988. Where do channels begin? Nature 336, 232-234.

Moody, J.A., 2010. Plot-scale sediment transport processes on a burned hillslope as a function of particle size. Proceedings of the 9th Federal Interagency Sedimentation Conference, June 27-July 1, 2010, Las Vegas, NV. (12 pp.). Moody, J.A., 2012. An analytical method for predicting postwildfire peak discharges. U.S. Geological Survey Scientific Investigations Report 2011–5236. (36 pp.).

Moody, J.A., Ebel, B.A., 2012a. Hyper-dry conditions provide new insights into the cause of extreme floods after wildfire. Catena 93. 58-63.

- Moody, J.A., Ebel, B.A., 2012b. Difference infiltrometer: a method to measure temporally variable infiltration rates during rainstorms. Hydrological Processes. http://dx.doi.org/ 10.1002/hyp.9424.
- Moody, J.A., Kinner, D.A., 2006. Spatial structures of stream and hillslope drainage networks following gully erosion after wildfire. Earth Surface Processes and Landforms 31, 319–337.
- Moody, J.A., Martin, D.A., 2001a. Post-fire, rainfall intensity-peak discharge relations for three mountainous watersheds in the western USA. Hydrological Processes 15, 2981–2993.
- Moody, J.A., Martin, D.A., 2001b. Initial hydrologic and geomorphic response following a wildfire in the Colorado Front Range. Earth Surface Processes and Landforms 26, 1049–1070.
- Moody, J.A., Martin, D.A., 2001c. Hydrologic and sedimentologic response of two burned watersheds in Colorado. U.S. Geological Water Resources Investigation Report 01–4122. (142 pp.).
- Moody, J.A., Martin, D.A., 2004. Wildfire impacts on reservoir sedimentation in the western United States. Proceedings of the Ninth International Symposium on River Sedimentation, October 18–21, 2004, Yichang, China, pp. 1095–1102.
- Moody, J.A., Martin, D.A., 2009a. Synthesis of sediment yields after wildland fire in different rainfall regimes in the western United States. International Journal of Wildland Fire 18, 96–115.
- Moody, J.A., Martin, D.A., 2009b. Forest fire effects on geomorphic processes. In: Robichaud, P.R., Cerdà, A. (Eds.), Fire Effects on Soils and Restoration Strategies. Science Publishers, New Hampshire, Enfield, NH, pp. 41–79.
- Moody, J.A., Nyman, P., 2012. Variations in soil erodibility with depth after wildfire. U.S. Geological Survey Scientific Investigation Report 2012–5233.
- Moody, J.A., Smith, J.D., Ragan, B.W., 2005. Critical shear stress for erosion of cohesive soils subjected to temperature typical of wildfires. Journal of Geophysical Research 110, F01004. http://dx.doi.org/10.1029/2004JF000141 (13 pp.).
- Moody, J.A., Martin, D.A., Oakley, T.M., Blanken, P.D., 2007. Temporal and spatial variability of soil temperature and soil moisture after a wildfire. U.S. Geological Survey, Scientific Investigations Report 2007–5015. (89 pp.).
- Moody, J.A., Martin, D.A., Haire, S.L., Kinner, D.A., 2008a. Linking runoff response to burn severity after wildfire. Hydrological Processes 22, 2063–2074.
- Moody, J.A., Martin, D.A., Cannon, S.A., 2008b. Post-wildfire erosion response in two geologic terrains in the western USA. Geomorphology 95, 103–118.
- Moody, J.A., Kinner, D.A., Úbeda, X., 2009. Linking hydraulic properties of fire affected soils to infiltration and water repellency. Journal of Hydrology 379, 291–303.
- Moreno, H.A., Vivoni, E.R., Gochis, D.J., 2012. Utility of quantitative precipitation estimates for high resolution hydrologic forecasts in mountain watershed of the Colorado Front Range. Journal of Hydrology 438–439. http://dx.doi.org/10.1016/ j.jhydrol.2012.03.019 (66–83).
- Moritz, M.A., Moody, T.J., Krawchuk, M.A., Hughes, M., Hall, A., 2010. Spatial variation in extreme winds predicts large wildfire locations in chaparral ecosystems. Geophysical Research Letters 37, L04801. http://dx.doi.org/10.1029/2009GL041735.
- Moss, A.J., 1988. Effects of flow-velocity variation on rain-driven transportation and the role of rain impact in the movement of solids. Australian Journal of Soil Research 26, 443–450.
- Murphy, B.P., Williamson, G.J., Bowman, D.M.J.S., 2011. Fire regimes: moving from a fuzzy concept to geographic entity. New Phytologist 192, 316–318.
- Nachabe, M.H., Illangasekare, T.H., Morel-Seytoux, H.J., Ahuja, L.R., 1997. Infiltration over heterogenous watersheds: influence of rain excess. Journal of Hydrologic Engineering 2 (3), 140–143.
- Nakagoshi, N., Nehira, K., Takahashi, F., 1987. The role of fire in pine forests of Japan. In: Trabaud, L. (Ed.), The Role of Fire in Ecological Systems. SPB Academic Publishing, The Hague, The Netherlands, pp. 91–119.
- Neary, D.G., 2011. Impacts of wildfire severity on hydraulic conductivity in forest, woodland and grassland soils. In: Elange, L. (Ed.), Hydraulic Conductivity–Issues, Determination, and Application. InTech Publishers, Rijeka, Croatia, pp. 123–142.
- Neary, D.G., Gottfried, G.J., 2002. Fires and floods: post-fire watershed responses. In: Viegas, V.R. (Ed.), Forest Fire Research & Wildland Fire Safety. Millpress, Rotterdam, pp. 1–9.
- Neary, D.G., Ryan, K.C., DeBano, L.F., 2005. Wildland fire in ecosystems, effects of fire on soil and water. U.S. Department of Agriculture, Forest Service Rocky Mountain Research Station, Fort Collins. General Technical Report RMRS-GTR-42-volume 4 (250 pp.).
- Nicks, A.D., Lane, L.J., Gander, G.A., 1995. Weather generator. In: Flanagan, D.C., Nearing, M.A. (Eds.), U.S. Department of Agriculture–Water Erosion Prediction Project Hillslope Profile and Watershed Model Documentation. NSER Report No. 10. U.S. Department of Agriculture, Agricultural Research Service, National Soil Erosion Research Laboratory, West Lafayette, IN, pp. 1.1–2.22.
- NIFC, 2011. National Interagency Fire Center, fire information—wildland fire statistics. online: http://www.nifc.gov/fire_info/lg_fires.htm (accessed 12 March 2011).
- Nikora, V., Goring, D., McEwan, I., Griffiths, G., 2001. Spatially averaged open-channel flow over rough bed. Journal of Hydraulic Engineering 127 (2), 123–133.
- Nimmo, J.R., Schmidt, K.M., Perkin, K.S., Stock, J.D., 2009. Rapid measurement of fieldsaturated hydraulic conductivity for areal characterization. Vadose Zone Journal 8, 142–149.
- Nishimune, N., Onodera, S., Naruoka, T., Birmano, M.D., 2003. Comparative study of bedload sediment yield processes in small mountainous catchments covered by secondary and disturbed forests, western Japan. Hydrobiologia 494, 265–270.
- NOAA, 2012. Precipitation Frequency Data Server. accessed October 2012 http:// dipper.nws.noaa.gov/hdsc/pfds/.

- Novák, V., Lichner, L., Zhang, B., Knava, K., 2009. The impact of heating on the hydraulic properties of soils sampled under different plant cover. Biologia 64, 483–486.
- NRCS, 1986. Urban hydrology for small watersheds. US Department of Agriculture, Natural Resources Conservation Services: Technical Release, 55 (159 pp.).
- Nyman, P., Sheridan, G., Lane, P.N.J., 2010. Synergistic effects of water repellency and macropore flow on the hydraulic conductivity of a burned forest soil, south-east Australia. Hydrological Processes 24, 2871–2887.
- Nyman, P., Sheridan, G., Smith, H.G., Lane, P.N.J., 2011. Evidence of debris flow occurrence after wildfire in upland catchments of south-east Australia. Geomorphology 125, 383–401.
- Ogden, F.L., Richardson, J.R., Julien, P.Y., 1995. Similarity in catchment response 2. Moving rainstorms. Water Resources Research 31 (6), 1543–1547.
- O'Donnell, A.J., Boer, M.M., McCaw, W.L., Grierson, P.F., 2011. Vegetation and landscape connectivity control wildfire intervals in unmanaged semi-arid shrublands and woodlands in Australia. Journal of Biogeography 38, 112–124.
- Onda, Y., Dietrich, W.E., Booker, F., 2008. Evolution of overland flow after a severe forest fire. Point Reyes, California. Catena 72, 13–20.
- Ormsbee, L.E., 1989. Rainfall disaggregation for continuous hydrologic modeling. ASCE Journal of Hydraulic Engineering 115 (4), 507–528.
- Overton, D.E., 1970. Route or convolute. Water Resources Research 6 (1), 43-52.
- Pannkuk, C.D., Robichaud, P.R., 2003. Effectiveness of needle cast at reducing erosion after forest fires. Water Resources Research 39 (12). http://dx.doi.org/ 10.1029/2003WR002318 (9 pp.).
- Parks, D.S., Cundy, T.W., 1989. Soil hydraulic characteristics of a small southwest Oregon watershed following high-intensity wildfires. U.S. Department of Agriculture, Forest Service General Technical Report. PSW-109. 63–67.
- Parsons, A., Robichaud, P.R., Lewis, S.A., Napper, C., Clark, J.T., 2010. Field guide for mapping post-fire soil burn severity. General Technical Report RMRS-GTR-243.USDA Forest Service, Fort Collins (49 pp.).
- Pausas, J.G., Fernández-Muñoz, S., 2011. Fire regimes changes in the Western Mediterranean Basin: form fuel-limited to drought-drive fire regime. Climatic Change 110, 215–226. http://dx.doi.org/10.1007/s10584-011-0060-6.
- Pausas, J.G., Llovet, J., Rodrigo, A., Vallejo, R., 2008. Are wildfires a disaster in the Mediterranean basin?—a review. International Journal of Wildland Fire 17, 713–723.
- Peckham, S.D., 2008. Geomorphometry and spatial hydrologic modeling. In: Hengl, T., Reuter, H.I. (Eds.), Geomorphometry: Concepts, Software and Applications. Chapter 22. Developments in Soil Science, vol. 33. Elsevier, pp. 377–393.
- Peckham, S.D., Hinzman, L., Nolan, M., 2004. The TopoFlow Hydrologic Model: a New Community Project, AGU Hydrology Days 2004, March 10–12, 2004. Colorado State University, Fort Collins, CO.
- Pereira, P., Cerdà, A., Úbeda, X., Mataix-Solera, J., Arcenegui, V., Zavala, L., 2013. Modeling the impacts of wildfire on ash thickness in a short-term period, land degradation and development. http://dx.doi.org/10.1002/ldr.2195 (in press).
- Pezzatti, G.B., Zumbrunnen, T., Bürgi, M., Ambrosetti, P., Conedera, M., 2011. Fire regime shifts as a consequence of fire policy and socio-economic development: an analysis based on the change point approach. Forest Policy and Economics. http://dx.doi.org/10.1016/j.forpol.2011.07.002.
- Philip, J.R., 1969. Theory of infiltration. Advances in Hydroscience, 5. Academic, San Diego, CA, pp. 215–296.
- Pierce, J., Meyer, G., 2008. Long-term fire history for alluvial fan sediments: the role of drought and climate variability, and implications for management of Rocky Mountain forests. International Journal of Wildland Fire 17, 84–95.
- Pierson, T.C., Costa, J.E., 1987. A rheologic classification of subaerial sediment-water flows. Reviews in Engineering Geology VII, 1–12.
- Pierson, F.B., Robichaud, P.R., Spaeth, K.E., 2001. Spatial and temporal effects of wildfire on the hydrology of a steep rangeland watershed. Hydrological Processes 15, 2905–2916.
- Pierson, F.B., Robichaud, P.R., Moffet, C.A., Spaeth, K.E., Williams, C.J., Hardegree, S.P., Clark, P.E., 2008. Soil water repellency and infiltration in coarse-textured soils of burned and unburned sagebrush ecosystems. Catena 74, 98–108.
- Pierson, F.B., Moffet, C.A., Williams, C.J., Hardegree, S.P., Clark, P.E., 2009. Prescribed-fire effects on rill and interrill runoff and erosion in a mountainous sagebrush landscape. Earth Surface Processes and Landforms 34, 193–203.
- Pietraszek, J.H., 2006. Controls of post-fire erosion at the hillslope scale, Colorado Front Range, Master Thesis. Colorado State University. 124 pp.
- Pilgrim, D.H. (Ed.), 1987. Australian Rainfall and Runoff: a Guide to Flood Estimation. Engineers Australia, Barton, ACT.
- Prosser, I., Williams, L., 1998. The effect of wildfire on runoff and erosion in native Eucalyptus forest. Hydrological Processes 12, 251–265.
- Rahardjo, H., Lee, T.T., Leong, E.C., Rezaur, R.B., 2005. Response of a residual soil slope to rainfall. Canadian Geotechnical Journal 42, 340–351.
- Randerson, J.T., Liu, H., Flanner, M.G., Chambers, D.D., Jin, Y., et al., 2006. The impact of boreal forest fire on climate warming. Science 314, 1130–1132.
 Rawls, W.J., Brakensiek, D.L., Saxton, K.E., 1982. Estimation of soil water properties.
- Rawls, W.J., Brakensiek, D.L., Saxton, K.E., 1982. Estimation of soil water properties. Transaction American Society Agricultural Engineers 25 (5), 1316–1330.
- Reaney, S.M., Bracken, L.J., Kirkby, M.J., 2007. Use of the connectivity of runoff model (CRUM) to investigate the influence of storm characteristics on runoff generation and connectivity in semi-arid areas. Hydrological Processes 21, 894–906.
- Reinhardt, E., 2012. Using FOFEM 5.0 to estimate tree mortality, fuel consumption, smoke production and soil heating from wildland fire. http://www.firelab.org/ science-applications/fire-fuel/111-fofem Accessed October 2012.
- Regalado, C.M., Ritter, A., 2009. A bimodal four parameter lognormal linear model of soil water repellency. Hydrological Processes 23, 881–892.
- Renard, K.G., 1985. Rainfall simulators and USDA erosion research: History, perspective, and future. In: Lane, L.J. (Ed.), Proceedings of the Rainfall Simulator Workshop, January 14–15, 1985, Tucson, Arizona, pp. 3–6.

- Renard, K.G., Foster, G.R., Weesies, G.A., McCool, D.K., Yoder, D.C., 1997. Predicting soil erosion by water: a guide to conservation planning with the Revised Universal Soil Loss Equation (RUSLE). U.S. Department of Agriculture, Agriculture Handbook No. 703 (404 pp.).
- Reneau, S.L., Kuyumjian, G.A., 2004. Rainfall-runoff relations in Pueblo Canyon, New Mexico, after the Cerro Grande Fire. Los Alamos National Laboratory Report LA-UR-04-8810. (31 pp.).
- Renschler, C.S., Mannaerts, C., Kiekkrüger, D., 1999. Evaluating spatial and temporal variability in soil erosion risk—rainfall erosivity and soil loss ratios in Andalusia, Spain. Catena 34, 209–225.
- Rice, R.M., 1982. Sedimentation in the chaparral: how do you handle unusual events? In: Swanson, F.J., Janda, R.J., Dunne, T., et al. (Eds.), Workshop on Sediment Budgets and Routing in Forested Drainage Basins: Proceedings: Portland, Oregon, U.S. Department of Agriculture, Forest Service, Pacific Northwest Forest and Range Experiment Station, General Technical Report PNW-141, pp. 39–49.
- Richards, L.A., 1931. Capillary conduction of liquids through porous mediums. Physics 1 (5), 318–333.
- Ries III, K.G., Crouse, M.Y., 2002. The national flood frequency program, version 3: a computer program for estimating magnitude and frequency of floods for ungaged sites. U.S. Geological Survey Water-Resources Investigations Report 02–4168. (42 pp.).
- Rinaldo, A., Marani, A., Rigon, R., 1991. Geomorphological dispersion. Water Resources Research 27 (4), 513–525.
- Robichaud, P.R., 2000. Fire effects on infiltration rates after prescribed fire in northern Rocky Mountain forests, USA. Journal of Hydrology 231–232, 220–229.
- Robichaud, P.R., Hungerford, R.D., 2000. Water repellency by laboratory burning of four northern Rocky Mountain forest soils. Journal of Hydrology 231–232, 207–219.
- Robichaud, P.R., Miller, S.M., 1999. Spatial interpolation and simulation of post-burn duff thickness after prescribed fire. International Journal of Wildland Fire 9 (2), 137–143.
- Robichaud, P.R., Lillybridge, T.R., Wagenbrenner, J.W., 2006a. Effects of postfire seeding and fertilizing on hillslope erosion in north-central Washington, USA. Catena 67, 56–67.
- Robichaud, P.R., Elliot, W.J., Pierson, F.B., Hall, D.E., Moffet, C.A., 2006b. Erosion Risk Management Tool (ERMIT)). Ver. 2006.01.18. [Homepage of ERMIT Erosion Risk Management Tool], [Online]. Available: http://forest.moscowfsl.wsu.edu/cgi-bin/ fswepp/ermit/ermit.pl (2008, March3).
- Robichaud, P.R., Elliot, W.J., Pierson, F.B., Hall, D.E., Moffet, C.A., Ashmun, L.E., 2007a. Predicting postfire erosion and mitigation effectiveness with a web-based probabilistic erosion model. Catena 71, 229–241.
- Robichaud, P.R., Lewis, S.A., Laes, D.Y.M., Hudak, A.T., Kokaly, R.F., Zamudio, J.A., 2007b. Postfire soil burn severity mapping with hyperspectral image unmixing. Remote Sensing of Environment 108, 467–480.
- Robichaud, P.R., Elliot, W.J., Pierson, F.B., Hall, D.E., Moffet, C.A., Ashmun, L.E., 2007c. Erosion Risk Management Tool (ERMiT) user manual (version 2006.01.18). General Technical Report RMRS-GTR-188.U.S. Department of Agriculture, Forest Service, Rocky Mountain Research Station, Fort Collins, CO (24 pp.).
- Robichaud, P.R., Wagenbrenner, J.W., Brown, R.E., Wohlgemuth, P.M., Beyers, J.L., 2008a. Evaluating the effectiveness of contour-felled log erosion barriers as a post-fire runoff and erosion mitigation treatment in the western United States. International Journal of Wildland Fire 17, 255–273.
- Robichaud, P.R., Pierson, F.B., Brown, R.E., Wagenbrenner, J.W., 2008b. Measuring effectiveness of three postfire hillslope erosion barrier treatments, western Montana, USA. Hydrological Processes 22, 159–170.
- Robichaud, P.R., Lewis, S.A., Ashmun, L.E., 2008c. New procedure for sampling infiltration to assess post-fire water repellency. U.S. Department of Agriculture, Forest Service Research Note RMRS-RN-33, U.S. Forest Service, W10506, pp. 1–14. http://dx.doi.org/10.1029/2009WR008314.
- Robichaud, P.R., Wagenbrenner, J.W., Brown, R.E., 2010. Rill erosion in natural and disturbed forests: 1. Measurements. Water Resources Research 46 (14 pp.).
- Robichaud, P.R., Lewis, S.A., Wagenbrenner, J.W., Ashmun, L.F., Brown, R.E., 2013a. Postfire mulching for runoff and erosion mitigation, Part I: Effectiveness at reducing hillslope erosion rates. Catena. http://dx.doi.org/10.1016/j.catena.2012.11.015.
- Robichaud, P.R., Wagenbrenner, J.W., Lewis, S.A., Ashmun, L.F., Brown, R.E., Wohlgemuth, P.M., 2013b. Post-fire mulching for runoff and erosion mitigation, Part II: Effectiveness at reducing runoff and sediment yields from small catchments. Catena. http:// dx.doi.org/10.1016/j.catena.2012.11.016.
- Robinson, J.S., Sivapalan, M., Snell, J.D., 1995. On the relative roles of hillslope processes, channel routing, and network geomorphology in the hydrologic response of natural catchments. Water Resources Research 31 (12), 3089–3101.
- Rodriguez-Iturbe, I., Valdes, J.B., 1979. The geomorphologic structure of hydrologic response. Water Resources Research 15 (6), 1409–1420.
- Roering, J.J., Gerber, M., 2005. Fire and the evolution of steep, soil-mantle landscapes. Geology 33, 349–352.
- Romme, W.H., Knight, D.H., 1981. Fire frequency and subalpine forest succession along a topographic gradient in Wyoming. Ecology 62 (2), 319–326.
- Rowe, P.B., 1948. Influence of woodland chaparral on water and soil in central California. California Forest and Range Experiment Station.U.S. Forest Service, University of California (70 pp.).
- Rowe, P.B., Countryman, C.M., Storey, H.C., 1954. Hydrologic Analysis Used to Determine Effects of Fire on Peak Discharge and Erosion Rates in Southern California Watersheds. USDA Forest Service, California Forest and Range Experimental Station (49 pp.).
- Rulli, M.C., Rosso, R., 2005. Modeling catchment erosion after wildfires in the San Gabriel Mountains of southern California. Geophysical Research Letters 32. http://dx.doi.org/10.1029/2005GL023635 (L19401).

- Rulli, M.C., Bozzi, S., Spada, M., Bocchiola, D., Rosso, R., 2006. Rainfall simulations of a fire disturbed Mediterranean area. Journal of Hydrology 327, 323–338.
- Ryan, K.C., 2002. Dynamic interactions between forest structure and fire behavior in boreal ecosystems. Silva Fennica 36 (1), 13–39.
 Saco, P.M., Kumar, P., 2002. Kinematic dispersion in stream networks. 1. Coupling
- Saco, P.M., Kumar, P., 2002. Kinematic dispersion in stream networks. 1. Coupling hydraulic and network geometry. Water Resources Research 38 (11), 26-1–26-14. http://dx.doi.org/10.1029/2001WR000695.
- Santi, P.M., deWolf, V.G., Higgins, J.E., Cannon, S.H., Gartner, J.E., 2008. Sources of debris flow material in burned areas. Geomorphology 96, 310–321.Sass, O., Heel, M., Leistner, I., Stöger, F., Wetzel, K.-F., Friedmann, A., 2012a. Disturbance,
- Sass, O., Heel, M., Leistner, I., Stöger, F., Wetzel, K.-F., Friedmann, A., 2012a. Disturbance, geomorphic processes and recovery of wildfire slopes in North Tyrol. Earth Surface Processes and Landforms 37, 883–894.
- Sass, O., Haas, F., Schimmer, C., Heel, M., Bremer, M., Stöger, F., Wetzel, K.-F., 2012b. Impact of forest fires on geomorphic processes in the Tyrolean Limestone Alps. Geografiska Annaler 94, 117–133.
- Scheuermann, A., 2008. Water content dynamics in unsaturated soils—results of experimental investigations in laboratory and in situ. In: Toll, D.G., Augarde, C.E., Gallipoli, D., Wheeler, S.J. (Eds.), Unsaturated Soils: Advances in Geo-Engineering. Taylor & Francis Group, London, pp. 197–203.Schmidt, K.M., Roering, J.J., Stock, J.D., Dietrich, W.E., Montgomery, D.R., Schaub, T.,
- Schmidt, K.M., Roering, J.J., Stock, J.D., Dietrich, W.E., Montgomery, D.R., Schaub, T., 2001. The variability of root cohesion as an influence on shallow landslide susceptibility in the Oregon Coast Range. Canadian Geotechnical Journal 38, 995–1024.
- Schmidt, K.M., Hanshaw, M.N., Howle, J.F., Kean, J.W., Staley, D.M., Stock, J.D., Bawden, G.W., 2011. Hydrologic conditions and terrestrial laser scanning of post-fire debris flows in the San Gabriel Mountains, CA, U.S.A Italian Journal of Engineering Geology and Environment. http://dx.doi.org/10.4408/IJEGE.2011-03.B-064.
- Schumm, S.A., 1973. Geomorphic thresholds and complex response of drainage systems. In: Morisawa, M. (Ed.), Fluvial Geomorphology, Proceedings Volume of the Fourth Annual Geomorphology Symposia Series Held at Binghamton, New York, State University of New York at Binghamton, pp. 299–310.
- Schumm, S.A., 1979. Geomorphic thresholds: the concept and its applications. Transactions of the Institute of British Geographers 4, 485–515.
- Schumm, S.A., 1980. Some applications of the concept of geomorphic thresholds. In: Coates, D.R., Vitek, J.D. (Eds.), Thresholds in Geomorphology. Allen & Unwin, London, pp. 473–485.
- Scott, D.F., van Wyk, D.B., 1992. The effects of fire on soil water repellency, catchment sediment yields and streamflow. In: van Wilgen, B.W., Richardson, D.M., Kruger, F.J., van Hesbergen, H.J. (Eds.), Fire in South African Mountain Fynbos. Springer-Verlag, New York, pp. 216–239.
- Shakesby, R.A., 2011. Post-wildfire soil erosion in the Mediterranean: review and future research directions. Earth-Science Reviews 105, 71–100.
- Shakesby, R.A., Doerr, S.H., 2006. Wildfire as a hydrological and geomorphological agent. Earth-Science Reviews 74, 269–307.
- Shakesby, R.A., Coelho, C.O.A., Ferreira, A.D., Terry, J.P., Walsh, R.P.D., 1993. Wildfire impacts on soil erosion and hydrology in wet Mediterranean forest, Portugal. International Journal of Wildland Fire 3, 95–110.
- Shakesby, R.A., Doerr, S.H., Walsh, R.P.D., 2000. The erosional impact of soil hydrophobicity: current problems and future research directions. Journal of Hydrology 231–232, 178–191.
- Shakesby, R.A., Wallbrink, P.J., Doerr, S.H., English, P.M., Chafer, C.J., Humphreys, G.S., Blake, W.H., Tomkins, K.M., 2007. Distinctiveness of wildfire effects on soil erosion in south-east Australian eucalypt forest assessed in a global context. Forest Ecology and Management 238, 347–364.
- Sheridan, G.J., Lane, P.N.J., Noske, P.J., 2007. Quantification of hillslope runoff and erosion processes before and after wildfire in a wet *Eucalyptus* forest. Journal of Hydrology 343, 12–28.
- Sherman, L.K., 1932. Stream-flow from rainfall by the unit-graph method. Engineering News-Record 108, 501–505.
- Shin, S.S., 2010. Response of runoff and erosion with vegetation recovery in differently treated hillslopes after forest fire, Korea. 8th International Symposium on Ecohydraulics, Seoul, Korea, September 2010.
- Silins, U., Stone, M., Emelko, M.B., Bladon, K.D., 2009. Sediment production following severe wildfire and post-fire salvage logging in the Rocky Mountain headwaters of the Oldman river basin, Alberta. Catena 79, 189–197.
- Šimùnek, J., Šejna, M., Saito, H., Sakai, M., van Genuchten, M.Th, 2008. The HYDRUS-1D software package for simulating the movement of water, heat, and multiple solutes in variably saturated media, version 4.08. HYDRUS Software Series, 3. Department of Environmental Sciences, University of California Riverside, Riverside, California, USA, p. 330 (2008).
- Skamarock, W.C., 2004. Evaluating mesoscale NWP models using kinetic energy spectra. Monthly Weather Review 132, 3019–3032.
- Smart, G.M., Duncon, M.J., Walsh, J.M., 2002. Relatively rough flow resistance equations. Journal of Hydraulic Engineering 568–578 (June).
- Smith, R.R., 1972. The infiltration envelope: results from a theoretical infiltrometer. Journal of Hydrology 17, 1–21.
- Smith, R.E., Goodrich, D.C., 2000. Model for rainfall excess patterns on randomly heterogeneous areas. Journal of Hydrologic Engineering 5, 355–362.
- Smith, R.E., Parlange, J.-Y., 1978. A parameter efficient hydrologic infiltration model. Water Resources Research 14, 533–538.
- Smith, M.B., Koren, V., Zhang, Z., Wang, D., 1999. Semi-distributed and lumped modeling approaches: case study of NEXRAD data application to large headwater basins in the Arkansas River basin. Abstract. Spring Meeting American Geophysical Union, Boston. MA.
- Smith, R.E., Smettem, K.R.J., Broadbridge, P., Woolhiser, D.A., 2002. Infiltration theory for hydrologic applications. Water Resources Monograph, 15. American Geophysical Union, Washington, D.C. (212 pp.).

- Smith, M.B., Koren, V.I., Zhang, Z., Reed, S.M., Pan, J.-J., Moreda, F., 2004. Runoff response to spatial variability in precipitation: an analysis of observed data. Journal of Hydrology 298, 267–286.
- Smith, H.G., Sheridan, G.J., Lane, P.N.J., Bren, L.J., 2011a. Wildfire and salvage harvesting effects on runoff generation and sediment exports from radiata pine and eucalypt forest catchments, south-eastern Australia. Forest Ecology and Management 261, 570–581.
- Smith, H.G., Sheridan, G.J., Nyman, P., Child, D.P., Lane, P.N.J., Hotchkis, M.A.C., Jacobsen, G.E., 2011b. Quantifying sources of fine sediment supplied to post-fire debris flows using fallout radionuclide tracers. Geomorphology 139–140, 403–415.
- Smith, H.G., Sheridan, G.J., Land, P.N.J., Noske, P.J., Heijnis, H., 2011c. Changes to sediment sources following wildfire in a forested upland catchment, southeastern Australia. Hydrological Processes 25, 2878–2889.
- Smith, H.G., Blake, W.H., Owens, P.N., 2012. Discriminating fine sediment sources and the application of sediment tracers in burned catchments: a review. Hydrological Processes. http://dx.doi.org/10.1002/hyp.9537.
- Soil Conservation Service, 1972. National Engineering Handbook, Section 4, Soil Conservation Service. U.S. Department of Agriculture, Washington, D.C.
- Spigel, K.M., Robichaud, P.R., 2007. First-year post-fire erosion rates in Bitterroot National Forest, Montana. Hydrological Processes 21, 988–1005.
- Spreen, W.C., 1947. A determination of the effect of topography upon precipitation. Eos, Transactions American Geophysical Union 28 (2), 285–290.
- Springer, E.P., Hawkins, R.H., 2005. Curve number and peak flow responses following the Cerro Grande Fire on a small watershed. In: Moglen, G.E. (Ed.), Proceedings: 2005 Watershed Management Conference–Managing Watersheds for Human and Natural Impacts: Engineering, Ecological, and Economic Challenges: 2005 July 19–22, 2005. Williamsburg, VA, American Society of Civil Engineers, pp. 459–470.
- Staley, D.M., Wasklewicz, T.A., Kean, J.W., 2010. Observations of drainage network change in a recently burned watershed using terrestrial laser scanning. Geophysical Research Abstracts 12, EGU2010-4849-1.
- Staley, D.M., Kean, J.W., Cannon, S.H., Schmidt, K.M., Laber, J.L., 2012. Objective definition of rainfall intensity-duration thresholds for the initiation of post-fire debris flows in southern California. Landslides. http://dx.doi.org/10.1007/s10346-012-0341-9.
- Stoof, C.R., Wesseling, J.G., Ritsema, C.J., 2010. Effects of fire and ash on soil water retention. Geoderma 159, 276–285.
- Stoof, C.R., Moore, D., Ritsema, C.J., Dekker, L.W., 2011a. Natural and fire-induced soil water repellency in a Portuguese shrubland. Soil Science Society of America Journal 75 (6), 2283–2295.
- Stoof, C.R., De Kort, A., Bishop, T.F.A., Moore, D., Wesseling, J.G., Ritsema, C.J., 2011b. How rock fragments and moisture affect soil temperatures during fire. Soil Science Society of America Journal 75 (3), 1133–1143.
- Stoof, C.R., Vervoort, R.W., Iwema, J., van den Elsen, E., Ferreira, A.J.D., Ritsema, C.J., 2012. Hydrological response of a small catchment burned by experimental fire. Hydrology and Earth System Sciences 16, 267–285.
- Suhr, J.L., Jarrett, A.R., Hoover, J.R., 1984. The effects of soil air entrapment on erosion. Transactions of the ASABE 27, 93–98.
- Sun, T., Meakin, P., Jøssang, T., 1994. A minimum energy dissipation model for drainage basins that explicitly differentiates between channel networks and hillslopes. Physica A 210, 24–47.
- Swanson, F.J., 1981. Fire and geomorphic processes. In: Mooney, H.A., Bonnicksen, T.M., Christensen, N.L., et al. (Eds.), Fire and Ecosystem Processes, U.S. Department of Agriculture Forest Service General Technical Report WO-26, pp. 401–420.
- Swetnam, T.W., Baisan, C.H., 2003. Tree-ring reconstructions of fire and climate history in the Sierra Nevada and southwestern United States. In: Veblen, T.T., Baker, W.L., Montenegro, G., Swetnam, T.W. (Eds.), Fire and Climatic Change in Temperate Ecosystems of the Western Americas, Chapter 6. Springer, New York, pp. 158–195.
- Swetnam, T.W., Baisan, C.H., Caprio, A.C., Brown, P.H., 1992. Fire history in a Mexican oak-pine woodland and adjacent montane conifer gallery forest in southeastern Arizona. Symposium on the Ecology and Management of Oak and Associated Woolands: Perspectives in the Southwestern United States and Northern Mexico. Sierra Vista, Arizona, April 27–30, 1992, pp. 165–173.
- Takahashi, T., Sawada, T., 1994. Bedload prediction in steep mountain rivers. In: Cotroneo, G.V., Rumer, R.R. (Eds.), Hydraulic Engineering '94, Proceedings of the 1994 Conference, 1994 August 1–5. American Society of Civil Engineers, Buffalo, New York, pp. 810–814.
- Tessler, N., 2012. Documentation and analysis of wildfire regimes on Mount Carmel and the Jerusalem hills. In: Maos, J.O., Charney, I. (Eds.), Special Issue of Horizons in Geography, 79–80, pp. 184–193.
- Tomkins, K.M., Humphreys, G.S., Gero, A.F., Shakesby, R.A., Doerr, S.H., Wallbrink, P.J., Blake, W.H., 2008. Postwildfire hydrological response in an El Nino-Southern Oscillation-dominated environment. Journal of Geophysical Research: Earth Surface 113, F02023.
- Tromp-van Meerveld, H.J., McDonnell, J.J., 2006. Threshold relations in subsurface storm flow: 2. The fill and spill hypothesis. Water Resources Research 42, W02411. http://dx.doi.org/10.1029/2004WR003800.
- Tryhorn, L., Lynh, A., Abramson, R., Parkyn, K., 2008. On the meteorological mechanism during postfire flash floods: a case study. Monthly Weather Review 136, 1778–1791.
- Tsuyuzaki, S., Kushida, K., Kodama, Y., 2009. Recovery of surface albedo and plant cover after wildfire in a *Picea mariana* forest in interior Alaska. Climatic Change 93, 517–525.
- Turner, R., Roberts, N., Jones, M.D., 2008. Climatic pacing of Mediterranean fire histories from lake sedimentary microcharcoal. Global and Planetary Change 63, 317–324.
- Ulery, A.L., Graham, R.C., Bowen, L.H., 1996. Forest fire effects on soil phyllosilicates in California. Soil Science Society of America Journal 60 (1), 309–315.

- Vaezi, A.R., Bahrami, H.A., Sadeghi, S.H.R., Mahdian, M.H., 2011. Developing a nomograph for estimating erodibility factor of calcareous soils in north west of Iran. International Journal of Geology 4 (5), 93–99.
- van Dine, D., Rodman, R., Jordan, P., Dupas, J., 2005. Kuskonook Creek, an example of a debris flow analysis. Landslides 2 (4), 257–265.
- van Rompaey, A.J.J., Vieillefont, V., Jones, R.J.A., Montanarella, L., Verstraeten, G., Bazzoffi, P., Dostal, T., Krasa, J., deVente, J., Poesen, J., 2003. Validation of soil erosion estimates at European scale. European Soil Bureau Research Report No. 13, EUR 20827 EN.Office for Official Publications of the European Communities, Luxembourg (26 pp.).
- van Wagtendonk, J.W., Root, R.R., Key, C.H., 2004. Comparison of AVIRIS and Landsat ETM+ detection capabilities for burn severity. Remote Sensing of Environment 92, 397–408.
- Vannière, B., Colombaroli, D., Chapron, E., Leroux, A., Tinner, W., Magny, M., 2008. Climate versus human-driven fire regimes in Mediterranean landscapes: the Holocene record of Lago dell'Accesa (Tuscany, Italy). Quaternary Science Reviews 27, 1181–1196.
- Veblen, T.T., Kitzberger, T., Donnegan, J., 2000. Climatic and human influences on fire regimes in ponderosa pine forests in the Colorado Front Range. Ecological Applications 10, 1178–1195.
- Veblen, T.T., Baker, W.L., Montenegro, G., Swetnam, T.W., 2003. Fire and Climate Change in Temperature Ecosystems of Western Americas. Springer-Verlag, New York (444 pp.).
- Venugopal, V., Foufoula-Georgiou, E., Sapozhnikov, V., 1999. Evidence of dynamic scaling in space-time rainfall. Journal of Geophysical Research 104 (D24), 31599–31610.
- Vivoni, E.R., Tai, K., Gochis, D.J., 2009. Effects of initial soil moisture on rainfall generation and subsequent hydrologic response during the North American monsoon. Journal of Hydrometeorology 10, 644–664.
- Wagenbrenner, J.W., MacDonald, L.H., Rough, D., 2006. Effectiveness of three post-fire rehabilitation treatments in the Colorado Front Range. Hydrological Processes 20, 2989–3006.
- Wagenbrenner, J.W., Robichaud, P.R., Elliot, W.J., 2010. Rill erosion in natural and disturbed forests: 2. Modeling approaches. Water Resources Research 46, W10507. http://dx.doi.org/10.1029/2009WR008315 (12 pp.).
- Wainwright, J., Parsons, A.J., 2002. The effect of temporal variations in rainfall on scale dependency in runoff coefficients. Water Resources Research 38 (12), 7-1–7-10. http://dx.doi.org/10.1029/2000WR000188.
- Wang, Z., Feyen, J., Nielsen, D.R., van Genuchten, M.T., 1997. Two-phase flow infiltration equations accounting for air entrapment effects. Water Resources Research 33, 2759–2767.
- Wang, Z., Feyen, J., van Genuchten, M.T., Nielsen, D.R., 1998. Air entrapment effects on infiltration rate and flow instability. Water Resources Research 34, 213–222.
- Weber, W.A., 1976. Rocky Mountain Flora, 5th edition. University Press of Colorado, Niwot, Colorado (479 pp.).
- Wells II, W.G., 1987. The effects of fire on the generation of debris flows in southern California. In: Costa, J.E. (Ed.), Debris Flow/Avalanches: Process, Recognition, and Mitigation, Reviews in Engineering Geology, III, pp. 105–114.
- Wendt, C.K., Beringer, J., Tapper, N.J., Hutley, L.B., 2007. Local boundary-layer development over burnt and unburnt tropical savanna: an observational study. Boundary-Layer Meteorology 124 (2), 291–304.
- Westerling, A.L., Gerchunov, A., Brown, T.J., Cayan, D.R., Dettinger, M.D., 2003. Climate and wildfire in the western United States. Bulletin of the American Meteorological Society 84 (5), 595–604.
- Westerling, A.L., Turner, M.G., Smithwick, E.A.H., Romme, W.H., Ryan, M.G., 2011. Continued warming could transform Greater Yellowstone fire regimes by mid-21st century. Proceedings of the National Academy of Sciences 108 (32), 13165–13170.
- Western, A.W., Blöschl, G., Grayson, R.B., 2001. Toward capturing hydrologically significant connectivity in spatial patterns. Water Resources Research 37 (1), 83–97.
- White, I., Sully, M.J., 1987. Macroscopic and microscopic capillary length and time scales from field infiltration. Water Resources Research 23 (8), 1514–1522.
- Whitlock, C., Marlon, J., Briles, C., Brunelle, A., Long, C., Bartleir, P.J., 2008. Long-term relations among fire, fuel, and climate in the northwestern US based on lakesediment studies. International Journal of Wildland Fire 17, 72–83.
- Whitlock, C., Higuera, P.E., McWethy, D.B., Briles, C.E., 2010a. Paleoecological perspectives on fire ecology: revisiting the fire-regime concept. The Open Ecology Journal 3, 6–23.
- Whitlock, C., Tinner, W., Newman, L., Kiefer, T., 2010b. Fire in the earth system: a paleoperspective. Pages News 18 (2), 55–95.
- Wiberg, P.L., Smith, J.D., 1987. Calculations of the critical shear stress for motion of uniform and heterogeneous sediment. Water Resources Research 23, 1471–1480.Williams, A.G., Moody, J.A., 2003. Historical Rainfall Characteristics Over the Hayman
- Burned Area. Report for Colorado Water Science Center (17 pp.). Winchell, M., Gupta, H.V., Sorooshian, S., 1998. On the simulation of infiltrationand saturation-excess runoff using radar-based rainfall estimates: effects of algorithm uncertainty and pixel aggregation. Water Resources Research 34 (10), 2655–2670.
- Wischmeier, W.H., Smith, D.D., 1978. Predicting rainfall erosion losses: a guide to conservation planning. Agriculture Handbook No. 537. U.S. Department of Agriculture, Washington D.C, pp. 13–27.
- Wischmeier, W.H., Johnson, C.B., Cross, B.V., 1971. A soil erodibility nomograph for farmland and construction sites. Journal of Soil and Water Conservation 26, 189–193.
- Wohl, E.E., 2010. Mountain Rivers Revisited, Water Resources Monograph, 19. American Geophysical Union (573 pp.).
- Wohlgemuth, P.M., Hubbert, K.R., 2008. The effects of fire on soil hydrologic properties and sediment fluxes in chaparral steeplands. Southern California, Riverside, CA, U.S. Department of Agriculture, Forest Service, General. Technical Report PSW-GTR-189. 115–122.

Wondzell, S.M., King, J.G., 2003. Postfire erosional processes in the Pacific Northwest and Rocky Mountain regions. Forest Ecology and Management 178, 75–87.

- Woods, S.W., Balfour, V.N., 2008. The effect of ash on runoff and erosion after a severe forest wildfire, Montana, USA. International Journal of Wildland Fire 17, 535–548. Woods, S.W., Balfour, V.N., 2010. The effect of soil texture and ash thickness on the post-
- fire hydrological response from ash-covered soils. Journal of Hydrology 393, 274–286.
 Woods, S.W., Birkas, A., Ahl, R., 2007. Spatial variability of soil hydrophobicity after wildfires in Montana and Colorado. Geomorphology 86, 465–479.
- Wu, W., Geller, M.A., Dickinson, R.E., 2002. The response of soil moisture to long-term variability of precipitation. Journal of Hydrometeorology 3 (5), 604–613.
- Xue, J., Gavin, K., 2008. Effect of rainfall intensity on infiltration into partly saturated slopes. Geotechnical Geological Engineering 26, 199–209.
- Yair, A., Raz-Yassif, N., 2004. Hydrological processes in a small arid catchment: scale effects of rainfall and slope length. Geomorphology 61, 155–169.
- Yang, C.T., 2006. Noncohesive sediment transport. Erosion and Sedimentation Manual, U.S. Department of the Interior, Bureau of Reclamation, Chapter 3. 1–104.
- Yates, D.N., Warner, T.T., Leavesley, G.H., 2000. Prediction of a flash flood in complex terrain. Part II: A comparison of flood discharge simulations using rainfall input from radar, a dynamic model, and an automated algorithmic system. Journal of Applied Meteorology 39, 815–825.
- Yu, B., Rose, C.W., Coughlan, K.J., Fentie, B., 1997. Plot-scale rainfall-runoff characteristics and modeling at six sites in Australia and southeast Asia. Transactions of the American Society of Agricultural Engineers 40 (5), 1295–1303.